

Revised as of January 1st, 2010 © Copyright, Real Property Association of Canada, 2009-



REALPAC Office Green Lease

National Standard Lease for Single-Building Projects 1.05 – 2021



About Real Property Association of Canada ("REALpac REALPAC")

REALpac is Canada's premier association for investment real property leaders. Its mission is to bring together the country's real property investment leaders to collectively influence public policy, to educate government and the public, and to ensure stable and beneficial real estate capital and property markets in Canada.

REALpac members currently own in excess of CDN \$150 Billion in real estate assets located in the major centres across Canada and around the world. REALpac members are comprised of real estate investment trusts, publicly traded and large private companies, banks, brokerages, crown corporations, investment dealers, life companies and pension funds. REALpac member assets include retail, office, industrial, hotel, multi-residential (apartments) and seniors housing.

Real Property Association of Canada

One University Avenue

Founded in 1970, REALPAC is the national leadership association dedicated to advancing the long-term vitality of Canada's real property sector. Our members include publicly traded real estate companies, real estate investment trusts (REITs), pension funds, private companies, fund managers, asset managers, developers, government real estate agencies, lenders, investment dealers, brokerages, consultants/data providers, large general contractors, and international members. Our members represent all asset classes in Canada – office, retail, industrial, apartment, hotel, seniors residential – from coast, to coast, to coast. Visit us at realpac.ca

REALPAC
77 King Street West
TD North Tower
Suite 14104030

Toronto, Ontario M5J 2PK 1H1 CANADA

Telephone: (416) 642-2700

Facsimile: (416) 642-2727

E-mail: clane info@realpac.ca

Web: www.realpac.ca

For Educational Purposes

The opinions and statements set forth herein reflect the viewpoint of Real Property Association of Canada ("REALpac")REALPAC at the time of publication but do not necessarily reflect the viewpoint of its Members. While a great deal of care has been taken to provide accurate and current information, neither REALpacREALPAC nor its editors or staff assume responsibility for the use of, effect of or appropriateness of the language, wording or standards contained in the Office GREENLEASE National Standard Lease for Single Building Projects (the "Lease"), or any typographical or printing errors or omissions in the Lease. Further, the information presented in the Lease is subject to applicable local, provincial and federal laws and regulations, court cases and any revisions of the same. This Lease is for educational purposes only with the understanding that REALpacREALPAC is not engaged in rendering legal, accounting or any other professional service.

GREENLEASE is a trade marktrademark of REALpacREALPAC.

Acknowledgements

REALPAC thanks Melissa McBain, Partner, Daoust Vukovich LLP for leading the 2020-2021 legal update of the GREENLEASE.

REALPAC also thanks its members who provided feedback in the revision process.

Publication Data

The Lease, version 1.01 <u>was</u> initially released as of June 1st, 2008, and version 1.02 <u>was</u> released as of March 30th, 2009. <u>Version 1.03</u> was released as of February 5th, 2010. Version 1.04 was released as of December 5th, 2017, and version 1.05 was released as of December 13th, 2021.

ISBN 978 0 9739934 2 4978-0-9739934-8-6

REALPAC is the owner of all copyright in the Lease. All rights reserved. No part of the Lease may be reproduced, transmitted or otherwise used in whole or in part in any form or by any means, except as expressly set out herein. Further, no person shall use the Lease, in whole or in part, in any form or by any means, to create any precedent for resale or license for remuneration.

Comments or Questions

If there are any comments with respect to this publication, please respond in writing to:

Carolyn Lane, Vice President, Research & Communications
REALpac
One University Avenue
REALPAC
77 King Street West
TD North Tower

<u>REALpac</u> <u>REALPAC</u> Office GREENLEASE National Standard Lease for Single-Building Projects <u>1.03</u>1.05 - <u>2010</u>2021

Suite 14104030 Toronto, Ontario M5J-2PK 1 H1 CANADA

Telephone: (416) 642-2700 Facsimile: (416) 642-2727 E-mail: clane info@realpac.ca

Register for Updates

User may send an email to <u>claneinfo</u>@realpac.ca to be notified of notes, bulletins and updates to this document and to ensure timely access to the latest authentic version for comparative purposes. <u>REALpac</u>REALPAC may make changes to the Lease at any time without notice.

When to Use the REALpac REALPAC Office GREENLEASE National Standard Lease for Single-Building Projects

The REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects was developed through a committee of lawyers, sustainability experts, and leasing specialists—and Leadership in Energy and Environmental Design ("LEED®") Accredited Professionals, based on research conducted around the world, and intended for use in all Canadian provinces outside Quebec. The base form of Lease contemplates a single office tower without a shared podium. It is suitable for the office tenancies only. In a mixed-use building (such as a building with ground floor retail), the Proportionate Share definition may merit review.

A significant update of the GREENLEASE occurred in 2021.

The "environmental sustainability" elements of this Lease contemplate a Landlord-centric lease structure, with the Landlord driving the environmental sustainability objectives, environmental sustainability decision making and compliance. However, the "environmental sustainability" elements of this Lease are easily transposable to a tenant-centric model, and can be negotiated into a "shared responsibility" model as well.

While the application of this Lease is intended for office projects, the "environmental sustainability" elements of this Lease, and particularly Schedule "E", are also transposable with appropriate modifications to retail, industrial and institutional leases.

The Lease - 1.01 - 2008 was initially released as of June 1st, 2008. Since that time, REALpacREALPAC received many comments and suggestions as to how the Lease could be improved, and many of those suggestions are reflected in the revisions contained in the current version of the Lease. As of 2021, other influences in the office leasing market, such as densification, the need to reduce carbon emissions, health and wellness programs, climate hazard assessment and resiliency initiatives, and health emergency and pandemic responsiveness, merited a further review of the lease provisions.

Because the Lease can become a legally binding document governing the rights and obligations of landlords and tenants, it is imperative that owners and managers review the applicability of this document on a building-by-building and province by province basis and make such changes as may be prudent in the circumstances, and as their professional advisors may recommend. In all cases, certain building specific information and provisions will need to be provided. The Lease has been prepared based generally on the law of common law Canadian provinces and primarily the Province of Ontario.

Fair Use Policy – Version <u>1.03</u>1.05 - <u>2010</u>2021

1. Thank you for using the <u>REALpaeREALPAC</u> Office GREENLEASE National Standard Lease for Single-Building Projects. It is intended as a Canadian national standard (excluding Quebec) for use in single building office premises.

- 2. Users should refer to the document as "REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects 1.031.05 20102021". The use of "REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects 1.031.05 20102021" is intended to demonstrate that the document is authentic, accurate and unamended with the exception only of variable data and supplementary pages containing additions or modifications clearly identified as such. The use of "REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects 1.031.05 20102021" is a representation that the fixed language of the original document has not been changed from its original form. Therefore, the parties only need to pay particular attention to the filled-in blanks and the supplementary pages.
- 3. If any modifications are made to the fixed language of the REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects 1.031.05 20102021, the document cannot be referred to as the "REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects 1.031.05 20102021". All modifications must be disclosed to the other parties involved in the transaction.
- 4. Users are permitted to use, modify, or duplicate the REALpaeREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects content for commercial or other public use without securing specific written authorization from REALpaeREALPAC, subject to the above, unless otherwise stipulated by REALpaeREALPAC from time to time.
- 5. Acknowledgement must be given to the REALpacREALPAC in any materials, presentations or papers associated with the REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects and any prior version.
- 6. User may send an email to <u>claneinfo</u>@realpac.ca to be notified of notes, bulletins and updates to this document and to ensure timely access to the latest authentic version for comparative purposes. <u>REALpacREALPAC</u> may make changes to this document at its sole discretion at any time without notice.
- 7. Users make use of the REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects at its own risk and there is no warranty express or implied as to fitness for purpose. REALpacREALPAC will not be liable for any damages, direct or other, as a consequence of using the REALpacREALPAC Office GREENLEASE National Standard Lease for Single-Building Projects or any of its content.
- 8. Any questions regarding the above should be directed to Carolyn Lane, Vice-President, Research & Communications, the Real Property Association of Canada, One University Avenue REALPAC, 77 King Street West, TD North Tower, Suite

<u>REALpac</u> <u>REALPAC</u> Office GREENLEASE National Standard Lease for Single-Building Projects <u>1.03</u>1.05 - <u>2010</u>2021

14104030, Toronto, Ontario, M5J-2PK 1H1, Canada, telephone (416) 642-2700, facsimile (416) 642-2727, e-mail clane info@realpac.ca.

REALPAC OFFICE GREENLEASE NATIONAL STANDARD LEASE FOR SINGLE BUILDING PROJECTS – VERSION 1.03-2010 1.05-2021 TABLE OF CONTENTS

| | | Page |
|------------|---|------------------|
| ARTICI F 1 | - BASIC TERMS | 1 |
| 1.1 | Basic Terms | |
| | | |
| ARTICLE 2 | - SPECIAL PROVISIONS | <u>+2</u> |
| 2.1 | <*> [NTD: Reference to Schedule "E" should be added as 2.1] Management Plan 2 | 1Sustainability |
| ARTICLE 3 | - DEFINITIONS AND INTERPRETATION | <u>12</u> |
| 3.1 | Definitions | |
| 3.2 | Entire Agreement, Amendments, Waiver | |
| 3.3 | Acceptance and Application of Rent | |
| 3.4 | General Rules of Interpretation | |
| 3.5 | Successors | 9 <u>14</u> |
| ARTICLE 4 | - GRANT AND TERM | 9 <u>14</u> |
| 4.1 | Term, Demise | <u>Ω</u> 14 |
| 4.2 | Acceptance | |
| 4.3 | Quiet Enjoyment | |
| ARTICLE 5 | - RENT | |
| 5.1 | Basic Rent | |
| 5.2 | Additional Rent | |
| 5.3 | Adjustment Due to Measurement | |
| 5.4 | Payment of Rent - General | |
| 5.5 | Payment of Additional Rent | |
| 5.6 | Rent Deposit | |
| 5.7 | Security Deposit | |
| 5.8 | Net Lease | <u>1218</u> |
| ARTICLE 6 | - OPERATING COSTS AND TAXES | <u>1218</u> |
| 6.1 | Property Taxes Payable by Landlord | |
| 6.2 | Property Taxes Payable by Tenant | |
| 6.3 | Business Taxes and Other Taxes of Tenant | |
| 6.4 | Assessment Appeals | 13 19 |
| 6.5 | Operating Costs | <u>1319</u> |
| 6.6 | Limitations on Operating Costs | <u>14</u> 22 |
| 6.7 | Adjustments of Operating Costs | |
| 6.8 | Reduction or Control of Operating Costs and Utility Consumption | <u>16</u> 24 |
| ARTICLE 7 | - HVAC, UTILITIES AND OTHER LANDLORD SERVICES. | <u>1624</u> |
| 7.1 | Heating, Ventilating, Air Conditioning, and Utility Consumption | <u>1624</u> |

REALPAC OFFICE GREENLEASE NATIONAL STANDARD LEASE FOR SINGLE BUILDING PROJECTS – VERSION 1.03-20101.05-2021 TABLE OF CONTENTS

(continued)

| | | rage |
|-------------|---|-------------------------|
| 7.2 | Electricity and Other Utilities | 16 25 |
| 7.3 | Special HVAC Services and Utilities and Excess Quantities | |
| 7.4 | Other Landlord Services | |
| 7.5 | Additional Services Provided by Landlord | |
| 7.6 | Telecommunications | |
| 7.7 | Signs and Premises Identification | 19 28 |
| ARTICLE | 8 - OPERATION, CONTROL AND MAINTENANCE BY LANDLO | RD 19 29 |
| 8.1 | Operation of the Building Project by Landlord | 19 29 |
| 8.2 | Control of the Building Project by Landlord | |
| 8.3 | Name of Building Project | |
| 8.4 | Maintenance and Repair by Landlord | 20 30 |
| 8.5 | Access by Landlord | 20 30 |
| 8.6 | Relocation | 20 <u>31</u> |
| 8.7 | Health Emergency | |
| 8.8 | Amenity Facility | 33 |
| ARTICLE | 9 - MAINTENANCE AND ALTERATIONS BY TENANT | 21 <u>33</u> |
| 9.1 | Maintenance and Repair by Tenant | 21 33 |
| 9.2 | Alterations by Tenant | |
| 9.3 | Removal of Improvements and Fixtures | 23 <u>35</u> |
| 9.4 | Liens | |
| 9.5 | Notice by Tenant | 24 <u>36</u> |
| ARTICLE | 10 - USE OF PREMISES | <u>24</u> 37 |
| 10.1 | Permitted Use | 24 37 |
| 10.2 | Compliance with Laws | 24 37 |
| 10.3 | <u> •</u> | |
| 10.4 | | |
| <u>10.5</u> | Density Limit | 38 |
| ARTICLE | 11 - INSURANCE, LIABILITY AND INDEMNITY | 25 <u>38</u> |
| 11.1 | Tenant's Insurance | 25 38 |
| 11.2 | Form of Tenant Policies | 26 39 |
| 11.3 | Certified Copies and Notice to Landlord | 26 40 |
| 11.4 | | |
| 11.5 | Insurance Risks | 26 41 |
| 11.6 | | |
| 11.7 | | |
| 11.8 | Indemnity by Tenant | <u>28</u> 43 |
| ARTICLE | 12 - ASSIGNMENT SUBLETTING AND OTHER TRANSFERS | 2843 |

REALPAC OFFICE GREENLEASE NATIONAL STANDARD LEASE FOR SINGLE BUILDING PROJECTS – VERSION 1.03-20101.05-2021 TABLE OF CONTENTS

(continued)

| | | Page |
|----------------|---|---------------------------------------|
| 12.1 | Transfers | 2843 |
| 12.2 | Tenant's Notice, Landlord's Right to Terminate | |
| 12.3 | Conditions of Transfer | |
| 12.4 | Corporate Records | |
| 12.5 | Permitted Transfers | |
| 12.6 | No Advertising | 31 <u>46</u> |
| 12.7 | Sales or Dispositions by Landlord | |
| ARTICLE | 13 - LANDLORD FINANCING AND STATUS CERTIFICATES | 31 <u>47</u> |
| 13.1 | Subordination and Postponement | <u>31</u> 47 |
| 13.2 | Attornment | |
| 13.3 | Status Certificates | <u>32</u> 48 |
| 13.4 | Reliance | <u>3248</u> |
| ARTICLE | 14 - DAMAGE, DESTRUCTION | 32 <u>48</u> |
| 14.1 | Damage to Premises | <u>32</u> 48 |
| 14.2 | Abatement | <u>33</u> 49 |
| 14.3 | Termination Rights | <u>33</u> 49 |
| 14.4 | Landlord's Rights on Rebuilding | <u>3349</u> |
| ARTICLE | 15 - DEFAULT AND REMEDIES | 33 <u>50</u> |
| 15.1 | Events of Default | <u>33</u> 50 |
| 15.2 | Remedies | <u>34</u> <u>51</u> |
| 15.3 | Distress | <u>35</u> <u>52</u> |
| 15.4 | Interest and Costs | |
| 15.5 | Remedies Cumulative | <u>35</u> <u>52</u> |
| ARTICLE | 16 - MISCELLANEOUS | 35 <u>52</u> |
| 16.1 | Relationship of Parties | <u>35</u> <u>52</u> |
| 16.2 | Consent Not to be Unreasonably Withheld | <u>35</u> <u>53</u> |
| 16.3 | Overholding | <u>3653</u> |
| 16.4 | Registration | |
| 16.5 | Unavoidable Delay | |
| 16.6 | Decisions of Experts | · · · · · · · · · · · · · · · · · · · |
| 16.7 | Notices | |
| 16.8 | Confidentiality, Personal Information | |
| 16.9 | Power, Capacity and Authority | |
| 16.10 | | |
| | E A BUILDING- SPECIFIC INFORMATION | |
| SCHEDUL | E B SKETCH SHOWING PREMISES | 4058 |

REALPAC OFFICE GREENLEASE NATIONAL STANDARD LEASE FOR SINGLE BUILDING PROJECTS – VERSION 1.03-20101.05-2021 TABLE OF CONTENTS

(continued)

| | Page |
|---|-----------------------|
| SCHEDULE C RULES AND REGULATIONS | . 41 <u>59</u> |
| SCHEDULE D INDEMNITY AGREEMENT | . 4 <mark>5</mark> 63 |
| SCHEDULE E ENVIRONMENTAL SUSTAINABILITY MANAGEMENT PLAN | . 48 66 |

REALPAC OFFICE GREENLEASE NATIONAL STANDARD LEASE FOR SINGLE BUILDING PROJECTS – 1.031.05 - 20102021

THIS LEASE, dated <*>, is made by the Landlord and the Tenant named in it who, in consideration of the rents, covenants and agreements contained in this Lease, covenant and agree as follows:

ARTICLE 1 - BASIC TERMS

| (a) | (i) | Landlord: | 1.1 | Basic T | erms |
|------------|-------------------------|-----------------------------|-----------|----------|----------------------|
| | (ii) | Address of Landlord: | | <*> | |
| (b) | (i) | | | Tenant: | <*> |
| | (ii) | Address of Tenant: | | <*> | |
| (c) | (i) | | | Indemni | fier: <*> |
| | (ii) | Address of Indemnifier: | | <*> | |
| | (iii) | Indemnity Provisions: | | See Sch | edule "D" |
| (d) | Build | ling: | | <*> | |
| (e) | Prem | ises: | | <*> | |
| (f) | Renta | able Area of Premises: | | <*> | |
| (g) | (i) | | | Term: | <*> |
| | (ii) | Commencement Date: | | <*> | |
| | (iii) | Expiry Date: | | <*> | |
| (h) | Fixtu | ring Period: | | <*> | |
| (i) | Basic Rent: | | | | |
| Lease Y | ear | (i) Per Sq. (ii Ft./Year | i) Per Ye | ear | (iii) Per Month |
| (j) (k) | (i) Secur | rity Deposit: | | —Rent De | eposit: <*> |

ARTICLE 2 - SPECIAL PROVISIONS

2.1 <*> [NTD: Reference to Schedule "E" should be added as 2.1] Sustainability Management Plan

The Sustainability Management Plan is attached to this Lease as Schedule "E" and forms an integral part of this Lease.

[NTD: Draftsperson to insert any other applicable deal-specific special provisions, such as (by way of example only): options to extend, leasehold improvement allowance, rent free period, exclusive covenant, and parking rights]

ARTICLE 3 - DEFINITIONS AND INTERPRETATION

3.1 Definitions

- (a) "Additional Rent" means all amounts in addition to Basic Rent payable by the Tenant to the Landlord or any other Person pursuant to this Lease, other than Rental Taxes.
- (b) "Alterations" has the meaning set out in Section 9.2.
- (c) "Amenity Facility" means any amenity as constructed and modified from time to time located in or on the Project which may be available for use by the Tenant as well as other tenants of the Project from time to time, on terms and conditions determined by the Landlord, including, without limitation, any conference rooms or facilities, daycare facilities, fitness areas and facilities, showers, lockers, bicycle parking, food service areas, facilities for service animals, and any other building areas, facilities and/or services designated as an amenity for the Project by the Landlord from time to time. "Amenity Facilities" has a corresponding meaning.
- (d) (e) "Applicable Laws" means all statutes, laws, by-laws, regulations, ordinances, guidelines, letters of instruction, orders and requirements of governmental or other public authorities having jurisdiction in force from time to time, including Environmental Laws.
- (e) (d) "Basic Rent" means the rent payable pursuant to Section 5.1.
- (f) (e) "Building" means the Lands and the building and all other structures, improvements, facilities and appurtenances that have been or will be constructed situated on the Lands (above, at or below grade), including the Building Systems and includes the Common Areas and Facilities, all as may be altered, expanded, reduced or reconstructed from time to time.
- (g) "Building Resilience Plan" means the plan of the Landlord to make modifications to the Project to increase its defences and adaptability against current day and

- anticipated Natural Hazards and/or the changing occurrence, frequency, and magnitude of certain hazards over time due to climate change.
- (h) "Building Resilience Plan Costs" means and refers to those costs incurred by the Landlord in furtherance of the Building Resilience Plan.
- <u>(i)</u> (f) "Building Systems" means at any time: (i) all heating, ventilating and airconditioning and other climate control systems, all systems, equipment and facilities required to maintain or improve air circulation, air quality and/or air cleanliness, and other systems, services, installations and facilities installed in or servicing the Building including, without limitation, the following systems, services, installations and facilities: elevators and escalators, mechanical (including HVAC), plumbing, sprinkler, drainage and sewage, electrical and other utilities, lighting, sprinkler, life safety (including fire prevention, communications, security and surveillance), computer (including environmental, security and lighting control), ice and snow melting, refuse removal, window washing and music; (ii) all machinery, appliances, equipment, apparatus, components, computer software and appurtenances forming part of or used for or in connection with any of such systems, services, installations and facilities including, but not limited to, boilers, motors, generators, fans, pumps, pipes, conduits, ducts, valves, wiring, meters and controls, and the structures and shafts housing and enclosing any of them; and (iii) all Landlord owned or controlled telecommunications facilities, pathways, installations and equipment; and (iv) all facilities and equipment used in energy management including back-up power and Electric Car Charging Stations.
- (j) "Business Day" means any day which is not a Saturday, Sunday or a day observed as a holiday under the Applicable Laws in the province in which the BuildingProject is situated.
- (k) "Business Hours" means the normal business hours determined by the Landlord, for the Building, which shall not be less than the hours set out in Part 2 of Schedule "A" on Business Days. for the Building as determined by the Landlord from time to time, which shall not be less than the hours set out in Part 2 of Schedule "A" on Business Days.
- (i) "Business Taxes" means all taxes, rates, duties, levies, assessments, licence fees and other charges in respect of the use or occupancy of, or any business carried on by, tenants or other occupants of the Building.
- "Capital Tax" means the amount determined by multiplying each of the "Applicable Rates" by the "Capital" and totalling the products. "Capital" is the amount of capital which the Landlord determines, without duplication, is invested from time to time by the Landlord, the owner(s) of the Building (including any interest in the Building), any company related to the Landlord or the owner(s) within the meaning of the *Income Tax Act* (Canada), or all of them, in doing all or any of: acquiring, developing, expanding, redeveloping and improving the Building. Capital will not be increased by any financing or re-financing except to

4

the extent that the proceeds are invested in doing all or any of the foregoing. "Applicable Rate" is the capital tax rate specified from time to time under any law which imposes a tax in respect of the capital of corporations and for greater certainty includes Large Corporations Tax levied under the *Income Tax Act* (Canada) as amended from time to time. Each Applicable Rate will be considered to be the rate that would apply if each of the Landlord, the owner(s) of the Building and the related companies referred to above were taxable corporations that employed no capital outside the province in which the Building is located.

- (l) (k) "Carbon Offset Costs" shall mean and refer to the cost of purchasing tradeable units, denominated in tonnes of carbon dioxide ("CO2"), or the CO2 equivalent using the global warming potential of other Greenhouse Gases, Carbon Offset Credits where the purchase of such tradeable units is necessary to ensure compliance of the Building Project with any required target Greenhouse Gas emission level or energy consumption level as prescribed by Applicable Law Laws, or performance targets as are set out in Schedule "E".
- (1) "Carbon Offset Credits" shall mean and refer to tradeable units, denominated in tonnes of CO₂ or other Greenhouse Gas, or the CO₂ equivalent using the global warming potential of other Greenhouse Gases, the tradeability of which may be permitted voluntarily in a given market or legislatively by a level of government, and which tradeable units may be created as a result of activities undertaken by either the Landlord or the Tenant which cause, directly or indirectly, measurable Greenhouse Gas emission reductions within or in respect of the Building and that have financial or exchange value in the regulatory or voluntary trading market.
- (m) "Carbon Offset Credits" shall refer to tradable, non-tangible instruments representing a unit of carbon dioxide-equivalent (CO2e), typically one tonne, that is reduced, avoided or sequestered by a project and is certified/verified to an internationally recognized carbon accounting standard or program such as Greene Climate, Gold Standard, Verified Carbon Standard (VCS), The Climate Action Reserve, the American Carbon Registry or other Canadian equivalent.²
- (n) "Carbon Tax" shall mean and refer to the aggregate of all taxes, rates, duties, levies, fees, charges and assessments whatsoever, imposed, assessed, levied, confirmed, rated or charged against or in respect of the consumption by the Landlord in or at the BuildingProject of electricity, natural gas, propane or any other fossil fuel used to produce energy, such as heat, light or electricity, for the BuildingProject or any part of it or levied in lieu thereof, and levied against the Landlord or the BuildingProject by any local, provincial or federal government or any agency thereof having jurisdiction.

¹ https://www.worldgbc.org/advancing-net-zero-whole-life-carbon

²https://www.cagbc.org/cagbcdocs/zerocarbon/v2/CaGBC_Zero_Carbon_Building_Standard_v2_Performance.pdf

- (o) (n) "CDS" has the meaning set out in Section 7.6($\frac{eb}{D}$).
- (p) "Change of Control" means, in the case of any corporation or partnership, the transfer or issue by sale, assignment, subscription, transmission on death, mortgage, charge, security interest, operation of law or otherwise, of any shares, voting rights or interest which would result in any change in the effective control of such corporation or partnership, unless such change occurs as a result of trading in the shares of a public corporation listed on a recognized stock exchange in Canada or the United States ("US").
- (q) (p) "Commencement Date" means the date set out in or determined pursuant to Section 1.1(g)(ii).
- (q) "Common Areas and Facilities" means those areas, facilities, improvements, (r) installations and equipment in or around the Building Project existing from time to time that: (i) are neither rented nor designated nor intended by the Landlord to be rented;; and (ii) are provided or designated from time to time by the Landlord for use in common by the Landlord, the Tenant, other tenants of the **Building** Project or their sublessees, agents, employees, customers, invitees or licensees, whether or not those areas are open to the general public or to all tenants of the Building Project including, without limitation, the Amenity Facility (if any), the Building Systems, entrances, lobbies, access and service corridors, stairways, indoor and outdoor walkways (both open and enclosed), malls, courts and arcades (both open and enclosed), public seating areas and facilities, public washrooms, indoor and outdoor landscaping and landscaped areas, passageways or tunnels leading to any public walkway or other facilities or to other buildings or concourses, mailrooms, electrical, telecommunications, cable, meter, valve, mechanical, storage and janitor rooms, telecommunication and electrical risers, shipping and receiving areas and loading docks, package or passenger pick-up areas, waste disposal or recycling facilities, parking facilities, driveways, laneways and ramps and sidewalks, parks and other municipal facilities for which the Landlord directly or indirectly is subject to obligations in its capacity as owner of the Building Project or an interest in it, all as may be altered, expanded, reduced, reconstructed or relocated from time to time.
- (s) (r) "Damage" shall have the meaning as set out in Section 14.1
- (t) (s) "Default Rate" means the lesser of: (i) the Prime Rate plus five percent per annum; and (ii) the maximum rate permitted by Applicable Laws, calculated and compounded monthly not in advance.
- (u) "Density Limit" has the meaning set out in Section 10.5.
- (v) (t) "Early Termination" has the meaning set out in Section 12.3(d).
- (w) "Electric Car Charging Stations" shall mean and refer to those electric car charging stations constructed and maintained by or on behalf of the Landlord in or on the Project, whether required by law or voluntarily provided.

- (x) "Electric Car Charging Station Costs" shall mean and refer to the costs of providing, operating, administering, maintaining, repairing and replacing the Electric Car Charging Stations including, without limitation, the consumption of electricity if not directly charged to the user of such charging station.
- (y) "Environmental Laws" shall mean all statutes, laws, by-laws, regulations, codes, orders, environmental penalties, tickets, notices, standards, guidelines, criteria, policies and directives, approvals, licences and permits now or at any time hereafter in effect, made or issued by any municipal, provincial or federal government, or by any department, agency, tribunal, board or office thereof, or any other agency or source whatsoever; (collectively, an "Authority"), regulating, relating to or imposing liability or standard of conduct concerning the natural or human environment (including air, land, surface water, groundwater, waste, real and personal property, moveable and immoveable property, sustainability, building operations, recycling or resource consumption), public or occupational health and safety and the manufacture, importation, handling, use, reuse, recycling, transportation, storage, disposal, elimination and treatment of a substance, hazardous or otherwise.
- (v) "Environmental Management Plan" shall mean and refer to those provisions set out in Schedule "E" attached hereto.
- (w) "Environmental Objectives" shall mean those objectives more particularly set out in Sections 1.2 and 1.3 of Schedule "E" attached hereto.
- (z) (x) "Event of Default" has the meaning set out in Section 15.1.
- (aa) (y) "Expert" means any architect, engineer, (or in the case of Schedule "E", a LEED Accredited Professional ("AP") engineer or a LEED AP architect) land surveyor, environmental sustainability consultant, energy auditor or other professional consultant appointed by the Landlord who, in the opinion of the Landlord, is qualified to perform the function for which he or she is they are retained.
- (bb) (z) "Expiry Date" means the date set out in or determined pursuant to Section 1.1(g)(iii).
- (cc) (aa) "Fiscal Year" means the fiscal period(s) as designated by the Landlord from time to time. The Landlord may have different Fiscal Years for any one or more of the components of Additional Rent.
- (dd) "Fixturing Period" means the period, if any, specified in Section 1.1(h) provided to the Tenant to perform its fixturing of the Premises.
- (ee) (ce) "Greenhouse Gases" shall mean any or all of CO₂, methane (CH₄), nitrous oxide (N₂O), Sulphur Hexafluoride (SF₆), Perfluoromethane (CF₄),

Perfluoroethane (C₂F⁶), Hydrofluorocarbons (HFC's)¹³, any substance designated as a greenhouse gas by Applicable Laws or any other substance that is the subject of reporting obligations under the Government of Canada's notice with respect to reporting of greenhouse gases released under the Canadian Environmental Protection Act, 1999 on February 1613, 20082021 in the Canada Gazette, part I, vol. 142155, no. 7⁴, as updated from time to time, or a successor obligation or any equivalent notice published by any provincial government, and "Greenhouse Gas" means any one of them.

- (ff) "Gross Floor Area" means the total property area, as measured between the exterior walls of the Building(s). This includes all finished areas inside the Building(s) including supporting areas.⁵
- (gg) "Hazard Adaptation" means having the Project adjust its design and/or operations to respond to and mitigate the impacts of possible Natural Hazards and hazardous events that may be experienced during the Term of this Lease and/or amplified due to climate change. Adaptation measures will improve the Project's overall Resilience.
- (hh) "Hazardous Event" shall mean any occurrence of a Natural Hazard that negatively impacts the Building Systems or occupants.
- (dd) "Hazardous Substance" means: (ai) any solid, liquid, gaseous or radioactive (ii) substance (including radiation) which, when it enters into or onto the Building Project, exists in or on the Building Project or is present in the water supplied to the **Building**Project, or when it is released into the environment from the Building Project or any part thereof or is entrained from one building to another building, or into the water or the natural environment, is likely to cause, at any time, material harm or degradation to any other property or any part thereof, or to the natural environmental or material risk to human health, and includes, without limitation, any flammables, explosives, radioactive materials, asbestos, lead paint, polychlorinated biphenyls ("PCBs"), fungal contaminants (including, without limitation, and by way of example, stachybotrys chartarum and other moulds), legionella and other similar bacteria, mercury and its compounds, dioxans and furans, chlordane ("DDT"), polychlorinated biphenyls, chlorofluorocarbons ("CFCs"), hydro-chlorofluorocarbons ("HCFCs"), volatile organic compounds ("VOCs"), urea formaldehyde foam insulation, radon gas, chemicals known to cause cancer or reproductive toxicity, pollutants, contaminants, hazardous wastes, toxic or noxious substances or related materials, petroleum and petroleum products; or (bii) any substance declared to be hazardous or toxic under any Environmental Laws or that does not meet any prescribed standard or criteria made

¹³ Including: HFC-23, HFC-32, HFC-41, HFC-43, HFC-125, HFC-134, HFC-134A, HFC-143, HFC-143A, HFC-152, HFC-227EA, HFC-236FA and HFC-245CA.

⁴ https://canadagazette.gc.ca/rp-pr/p1/2021/202<u>1-02-13/html/sup1-eng.html</u>

⁵ https://www.nrcan.gc.ca/energy-efficiency/energy-star-canada/benchmarking-frequently-asked-questions/3787

- under any Environmental Laws now or hereafter enacted or promulgated by any Authority, or (c) both (a) and (b).
- (jj) "Health Emergency" means a situation in which the Landlord determines, based on advice from a medical professional, or a directive, bulletin, guideline, letter of instruction, notice or other form of communication from a public health authority, that occupants, tenants, invitees or contractors working in or on the BuildingProject are or may be exposed to imminent danger from a disease, virus or other biological or physical agents that may be detrimental to human health including, by way of example, only, Coronaviruses (including Severe Acute Respiratory Syndrome ("SARS") and 2019 novel coronavirus (2019-nCoV) ("COVID-19")), Legionnaires' disease, and Influenzavirus A strains and subtypes such as Avian Flu (H5N1).
- (kk) (ff) "Health Emergency Plan" shall mean and refer to a plan prepared by or for the Landlord for operating and managing the BuildingProject in response to a Health Emergency as it may be amended from time to time.
- (II) "Health Emergency Plan Costs" means the costs of preparing a pandemic risk assessment and/or a Health Emergency Plan, as well as the actual costs in dealing with a Health Emergency including implementing a Health Emergency Plan.
- (mm) (gg) "HVAC" shall mean heating, ventilation and air conditioning equipment.
- (nn) (hh) "Indemnifier" means the party named in Section 1.1(c)(i), if any.
- (oo) (ii) "Landlord" means the party named in Section 1.1(a)(i).
- (pp) "Landlord Data" shall mean any and all data collected by or on behalf of the Landlord in respect of the Common Areas and Facilities (including, without limitation, the Building Systems) and operation thereof, including occupant density and location, maintenance and repair, janitorial, security, utility consumption, Building services usage, environmental indicators, health and wellness indicators, hazard indicators, and all other such data which may be used by the Landlord from time to time in managing and operating the Project and the tenancies within the Project.
- (qq) (jj) "Lands" means the lands described in Part 1 of Schedule "A" (or such portion thereof as may be designated by the Landlord from time to time) and includes any external Common Areas and Facilities (if applicable) thereon, as such lands may be altered, expanded or reduced from time to time.
- (<u>rr</u>) (<u>kk</u>) "**Lease**" means this lease, including all Schedules, as it may be amended.
- (ss) (II) "Lease Year" means: (i) in the case of the first Lease Year, the period beginning on the Commencement Date and ending, as determined by the Landlord (and which may vary for different components of Rent), ending on either; (A) December 31st of the calendar year in which the Commencement Date occurred; or (B) on the last

day of the l2th consecutive full month after the expiry of the calendar month in which the Commencement Date occurs (except that if the Commencement Date occurs on the first day of a calendar month, the first Lease Year shall end on the day prior to the first anniversary of the Commencement Date); and; (ii) in the case of each subsequent Lease Year, consecutive 12 month periods, provided that the final Lease Year shall end on the Expiry Date.

- (tt) (mm) "Leasehold Improvements" means all Alterations, fixtures and improvements in or serving the Premises made from time to time by or on behalf of the Tenant or any prior occupant of the Premises including, without limitation, internal stairways, doors, hardware, partitions (excluding moveable partitions), lighting fixtures, Building standard window coverings and wall-to-wall carpeting (excluding carpeting laid over a finished floor and removable without damage to such floor), kitchenette millwork and built in dishwasher or similar built in appliances (if any), but excluding trade fixtures and furniture and equipment not of the nature of fixtures.
- (uu) "Measurement Standards" means, subject to Section 5.3 of this Lease, the Building Owners and Managers Association ("BOMA") International Measurement Standards which are identified in Part 3 of Schedule "A", provided that notwithstanding the foregoing or anything else contained in this Lease, the Landlord may, at its option from time to time, choose to measure the area of the Premises or any space included in the Building in accordance with thea more recent BOMA standard method of measurement then in effect from time to timeor a comparable alternative system in use in the commercial real estate industry.
- (vv) (oo) "Mortgage" means any mortgage, charge or security instrument (including a deed of trust or mortgage securing bonds) and all extensions, renewals, modifications, consolidations and replacements of any such item which may now or hereafter affect the Building or Project any part of it.
- (ww) (pp) "Mortgagee" means the mortgagee, chargee or other secured party (including a trustee for bondholders), as the case may be, who from time to time holds a Mortgage.
- (xx) "Natural Hazard" means and refers to any naturally occurring event, or those amplified by climate change, that may impact the Building Systems or occupants, such as flood (fluvial, pluvial, or storm surge), wildfire, extreme heat, extreme cold, hurricanes, tornados, hail and ice storms, earthquakes, tsunamis, high winds, sealevel rise, coastal flooding, drought and infestations.
- (yy) "Net Zero Carbon Building" means a building that has eliminated its operational carbon emissions (Scope 1 and Scope 2), and embodied carbon emissions (Scope 3) from any newly added materials associated with the construction process of a tenant fit-out or improvement, or as part of any interior or exterior maintenance of the existing Project, provided that, after all reduction efforts have been made to eliminate such emissions, the balance of remaining emissions is offset by Carbon

- Offset Credits or onsite, or procured, carbon-free renewable energy through Renewable Energy Credits or Certificates (RECs).⁶
- (zz) (qq) "Notice" has the meaning set out in Section 16.7.
- (aaa) (rr) "Operating Costs" has the meaning set out in Section 6.5.
- (bbb) (ss) "Permitted Transferee" means any entity which is an affiliate (as that term is defined as of the date of this Lease in the *Canada Business Corporations Act*) of the original named Tenant, and only for so long as it remains an affiliate of such original named Tenant.
- (ccc) (tt) "Person" means any individual, partnership, corporation, trust, trustee or other entity or any combination of them, and "Persons" means more than one Person.
- (ddd) "Plug Load Maximum" means the maximum amount of energy a tenant's equipment can use in or from its premises given the capacity of the Building Systems, or based on an annual energy budget as agreed upon by the Landlord and the Tenant.
- (eee) (uu) "Premises" means that part of the Building identified in Section 1.1(e) and approximately shown cross-hatched on Schedule "B", extending to: (i) the interior face of all exterior walls, doors and windows; (ii) the interior face of all interior walls, doors and windows separating the Premises from Common Areas and Facilities; (iii) the centre line of all interior walls separating the Premises from adjoining leaseable premises; and (iv) the top surface of the structural subfloor and the top surface of the suspended or plaster ceiling (or the bottom surface of the structural ceiling if there is no suspended or plaster ceiling). Any Building Systems located in the Premises do not form part of the Premises.
- (fff) (vv) "Prime Rate" means the annual rate of interest announced from time to time by the Canadian chartered bank chosen by the Landlord as the daily rate of interest used by such bank as a reference rate in setting rates of interest for Canadian dollar commercial loans and commonly referred to by such bank as its Canadian "prime rate".
- (ggg) (ww) "Project" means the Building and the Lands.
- (hhh) (xx) "Property Taxes" means the aggregate of all taxes, rates, duties, levies, fees, charges (including local improvement charges) and assessments whatsoever, imposed, assessed, levied, confirmed, rated or charged against or in respect of the BuildingProject or any part of the BuildingProject or any fixtures, equipment or improvements thereto from time to time by any lawful taxing or assessing authority, whether school, transit, municipal, regional, provincial, federal, parking, utilities authority, utility or otherwise, including extraordinary and special assessments,

⁶https://www.cagbc.org/cagbcdocs/zerocarbon/v2/CaGBC_Zero_Carbon_Building_Standard_v2_Performance.pdf

parking taxes, special transit or affordable housing surcharges collected through the property tax system, and any taxes or other amounts which are imposed in lieu of, or in addition to, any of the foregoing whether or not in existence on the Commencement Date and whether of the foregoing character or not, or against Landlord in respect of any of the same or in respect of any rental or other compensation receivable by Landlord and/or the owners of the Project in respect of the same, and any Taxestaxes on real property rents or receipts of such (as opposed to a tax on such rents as part of the income of Landlord), any Taxes taxes based, in whole or in part, upon the value of the Project, any commercial concentration or similar levy in respect of the Project excluding taxes on the income or profits of the Landlord except to the extent that they are levied in lieu of the foregoing. For clarification, Property Taxes shall not include any taxes personal to Landlord such as income tax, inheritance tax, gift tax or estate tax nor shall Property Tax include any penalties or fines incurred as a result of a Landlord's late payment of same, provided Tenant has in fact remitted such Property Taxes as and when required hereby and provided same are not being bona fide contested and/or withheld by Landlord.

- (iii) (yy) "**Proportionate Share**" means a fraction which has: (i) as its numerator, the Rentable Area of the Premises, and (ii) as its denominator, the Rentable Area of the Building.
- (jjj) "Renewable Energy Credits or Certificates (RECs)" means market instruments that represent the environmental benefits associated with one megawatt hour of electricity generated from renewable resources such as solar and wind (which can be purchased from a third party), provided that all RECs used must be certified by EcoLogo^M, Green-e or Canadian equivalent and generated from green power facilities in Canada.⁷
- (kkk) (zz) "Rent" means all Basic Rent and Additional Rent.
- (III) (aaa) "Rent Deposit" means any amount specified in Section 1.1(j)(i).

(mmm)(bbb)—"Rentable Area" means: (i) in the case of the Premises and any other premises included in the Building, the area of all floors of such premises determined in accordance with the Measurement Standards; and (ii) in the case of the Building the aggregate of the area of all premises in the Building that are rented, or designated or intended by the Landlord to be rented (whether actually rented or not) but excluding storage areas, basement and mezzanine areas (if any), determined in accordance with the Measurement Standards. The Rentable Area of the Building may be adjusted from time to time to reflect any alteration, expansion, reduction, recalculation or other change.

⁷https://www.cagbc.org/cagbcdocs/zerocarbon/v2/CaGBC_Zero_Carbon_Building_Standard_v2_Performance.pdf

- (nnn) (ccc) "Rental Taxes" means all goods and services, business transfer, multi-stage sales, sales, use, consumption, value-added or other similar taxes imposed by any federal, provincial or municipal government upon either the Landlord or the Tenant in respect of thethis Lease which is measured by or based in whole or in part directly upon the Rent payable under this Lease or in respect of the rental or rental value of premises under this Lease or the payments made by Tenant hereunder or the goods and services provided by Landlord hereunder including without limitation, the rental of the Premises and the provision of administrative services to Tenant hereunder, whether existing at the date of this Lease or hereafter imposed by any governmental authority including, without limitation, goods and services tax, harmonized sales tax, value added tax, business transfer tax, sales tax, federal sales tax, excise taxes or duties or any tax similar to the foregoing.
- (ooo) (ddd) "Repair Standard" shall have the meaning as set out in Section 8.4.
- (ppp) (eee) "Required Conditions" means that:
 - (i) the Tenant is the original named Tenant or a Permitted Transferee, has not undergone a Change of Control and is itself in occupation of and carrying on business from the whole of the Premises; and
 - (ii) the Tenant has paid all Basic Rent and Additional Rent as and when due and has not been in persistent default and is not in material default under this Lease.
- (qqq) "Resilience" means preparation for changing conditions and short-term responses to disruptive shocks (such as, by way of example, Natural Hazards including wildfire and flood events) and chronic stresses (such as, by way of example, changing heating and cooling degree days or precipitation levels) so that the Project can better withstand a Hazardous Event and recover more quickly afterwards.⁸
- (rrr) "Rules and Regulations" means the Rules and Regulations annexed hereto as Schedule "C" together with any amendments, deletions and additions made by the Landlord from time to time pursuant to Section 10.4, all of which shall form part of this Lease.
- (sss) (ggg) "Security Deposit" means any amount specified in Section $1.1(\frac{j)(iik}{j})$.
- (ttt) "Sustainability Management Plan" shall mean and refer to those provisions set out in Schedule "E" attached hereto.
- (uuu) "Sustainability Objectives" shall mean those objectives more particularly set out in Sections 1.2 and 1.3 of Schedule "E" attached hereto.
- (vvv) (hhh) "Tenant" means the party named in Section 1.1(b)(i).

 $[\]frac{8 https://documents.gresb.com/generated_files/survey_modules/2020/resilience/reference_guide/complete.html \#RS_definitions$

- (www) (iii) "Tenant Construction Manual" shall mean that document (if any) prepared by the Landlord in respect of the BuildingProject which sets out rules, specifications, and procedures for the design and construction of improvements and Alterations in and to the Premises and elsewhere by the Tenant, as may be specified in such manual.
- (xxx) (jjj)—"Tenant Procurement Guidelines" shall mean that document (if any) prepared by the Landlord in respect of the BuildingProject which sets out suggested specifications of Tenant furniture, fixtures, materials and equipment used or to be used by the Tenant in or on the BuildingProject or within the Tenant's Premises.
- (yyy) (kkk)—"Term" means the period of time specified in Section 1.1(g)(i) which commences on the Commencement Date and expires on the Expiry Date, unless terminated earlier pursuant to the provisions of this Lease.
- (III) "Transfer" means all or any of the following, whether by conveyance, written agreement or otherwise: (i) an assignment of this Lease in whole or in part; (ii) a sublease of all or any part of the Premises; (iii) the sharing or transfer of any right of use or occupancy of all or any part of the Premises regardless of duration; (iv) any mortgage, charge or encumbrance of this Lease or the Premises or any part of the Premises or other arrangement under which either this Lease or the Premises become security for any indebtedness or other obligation; and (v) a Change of Control, and includes any transaction or occurrence whatsoever (including, but not limited to, expropriation, receivership proceedings, seizure by legal process and transfer by operation of law), which has changed or might change the identity of the Person having use or occupancy of any part of the Premises.
- (aaaa) (mmm) "Transferee" means the Person to whom a Transfer is or is to be made.
- (bbbb) (nnn) "TSP" has the meaningmeans a telecommunications service provider as set out in Section 7.6(ba).
- (ccc) (ooo) "Unavoidable Delay" has the meaning set out in Section 16.5.

3.2 Entire Agreement, Amendments, Waiver

This Lease contains the entire agreement between the parties with respect to the subject matter of this Lease and there are no other agreements, promises or understandings, oral or written, between the parties in respect of this subject matter. This Lease may be amended only by written agreement between the Landlord and the Tenant. No electronic communications between the parties will have the effect of amending this Lease. No provisions of this Lease shall be deemed to have been waived by the Landlord or the Tenant unless such waiver is in writing signed by such party. If either the Landlord or the Tenant excuses or condones any default by the other of any obligation under this Lease, no waiver of such obligation shall be implied in respect of any continuing or subsequent default. The Landlord's receipt of Rent with knowledge of a breach shall not be deemed a waiver of any breach.

3.3 Acceptance and Application of Rent

Any endorsement, statement, condition, direction or other communication on or accompanying any Rent payment shall not be binding on the Landlord and the acceptance of any such payment shall be without prejudice to the Landlord's right to recover the balance of Rent then owing or to pursue any other remedy available to the Landlord. Any payment received by the Landlord may be applied towards amounts then outstanding under this Lease in such manner as the Landlord determines.

14

3.4 General Rules of Interpretation

- (a) Obligations as Covenants: Each obligation of the Landlord and the Tenant in this Lease shall be considered a covenant for all purposes. If the Tenant has failed to perform any of its obligations under this Lease, such obligations shall survive the expiration or other termination of this Lease.
- (b) Time: Time is of the essence of this Lease.
- (c) <u>Number, Gender</u>: The grammatical changes required to make the provisions of this Lease apply in the plural sense where the Tenant comprises more than one Person and to individuals (male or female), partnerships, corporations, trusts or trustees will be assumed as though in each case fully expressed.
- (d) Liability of Tenant: If the Tenant consists of more than one Person, the covenants of the Tenant shall be joint and several covenants of each such Person. If the Tenant is a partnership and not a limited liability partnership, each Person who is presently a partner of the partnership and each Person who becomes a member of any successor partnership shall be and continue to be bound jointly and severally for the performance of and shall be and continue to be subject to all of the terms, obligations and conditions of this Lease, whether or not such Person ceases to be a member of such partnership or successor partnership and whether or not such partnership continues to exist. If the Tenant is a limited liability partnership, each Person who is presently a partner of the limited liability partnership and each Person who becomes a member of any successor limited liability partnership shall be and continue to be bound jointly and severally for the performance of and shall be and continue to be subject to all of the terms, obligations and conditions of this Lease, whether or not such Person ceases to be a member of such limited liability partnership or successor limited liability partnership and whether or not such limited liability partnership continues to exist, to the full extent permitted by Applicable **Law**Laws.
- (e) <u>Governing Law</u>: This Lease shall be governed by and construed under the Applicable Laws of the jurisdiction in which the <u>BuildingProject</u> is located and the parties attorn and submit to the jurisdiction of the courts of such jurisdiction.
- (f) <u>Headings</u>: The headings of the Articles and Sections are included for convenience only, and shall have no effect upon the construction or interpretation of this Lease.

- (g) <u>Landlord as Trustee</u>: Any and all exculpatory provisions, releases and indemnities included in this Lease for the benefit of the Landlord are intended also to benefit the Mortgagees, any owner or lessor with an interest in the <u>BuildingProject</u> prior to the Landlord and property managers of the Landlord and the officers, directors, shareholders, employees, agents of each one of them and, for the purposes of such provisions, the Landlord is acting as agent or trustee on behalf of and for the benefit of the Persons mentioned above.
- (h) <u>Severability</u>: Should any provision of this Lease be or become invalid, void, illegal or not enforceable, such provision shall be considered separate and severable from this Lease and the remaining provisions shall remain in force and be binding upon the parties hereto as though such provision had not been included.

3.5 Successors

This Lease and everything herein contained shall extend to and bind the successors and assigns of the Landlord and the legal representatives, heirs, executors, administrators, successors and permitted assigns of the Tenant (as the case may be), except as may be specifically excepted herein.

ARTICLE 4 - GRANT AND TERM

4.1 Term, Demise

The Landlord hereby demises and leases the Premises to the Tenant for the Term (unless terminated earlier pursuant to this Lease), to have and to hold during the Term, subject to the terms and conditions of this Lease. The Landlord grants to the Tenant a non-exclusive licence throughout the Term to the benefit or use (as may be appropriate) of those Common Areas and Facilities which provide access to the Premises or which are generally made available to all tenants inof the BuildingProject, in common with other tenants of the BuildingProject and with all others entitled thereto, subject to the terms and conditions of this Lease.

4.2 Acceptance

The Tenant hereby leases and accepts the Premises from the Landlord and covenants to pay the Rent and to observe and perform all the covenants and obligations to be observed and performed by the Tenant pursuant to this Lease. The Tenant agrees that, except as may be specifically set out herein, the Premises are accepted on an "as is" basis and there is no promise, representation or undertaking binding upon the Landlord with respect to any alteration, remodelling or decoration of the Premises or with respect to the installation of equipment or fixtures in the Premises.

4.3 Quiet Enjoyment

If the Tenant pays the Rent, fully performs all its obligations under this Lease and there has been no Event of Default, then the Tenant shall be entitled, subject to the provisions of this Lease and to Applicable <u>LawLaws</u>, to peaceful and quiet enjoyment of the Premises for the Term, subject to the provisions of Section 8.7 and Section 16.5.

ARTICLE 5 - RENT

5.1 Basic Rent

The Tenant shall pay to the Landlord Basic Rent in the amount set out in Section 1.1(i) for the respective Lease Year, by equal consecutive monthly instalments in advance on the first day of each month, subject to any adjustment pursuant to Section 5.3.

5.2 Additional Rent

The Tenant shall also pay throughout the Term, at the times and in the manner provided in this Lease, all Additional Rent which shall, except as otherwise provided in this Lease, be payable within 15 days of receipt by the Tenant of an invoice, statement or demand for it.

5.3 Adjustment Due to Measurement

The Landlord may, from time to time, at its option, cause the Rentable Area of the Premises to be measured by an Expert in accordance with the Measurement Standards and deliver a certificate of measurement to the Tenant, and, if necessary as a result of such measurement, the annual Basic Rent and the calculation of Additional Rent shall be adjusted by the Landlord. The effective date of any such adjustment shall be:

- (a) in the case of any measurement made prior to or within six months of the Commencement Date, the date the Tenant is allowed possession of the Premises under this Lease; and
- (b) in all other cases, the date of the determination of the measurement.

Any such measurement by an Expert shall be final and binding on the Landlord and the Tenant. Neither the Landlord nor the Tenant may claim any adjustment to the annual Basic Rent or to the calculation of Additional Rent based on the Rentable Area of the Premises except in accordance with a measurement by an Expert made pursuant to this Section and, for greater certainty, neither the Landlord nor the Tenant may claim any adjustment to the annual Basic Rent or to the calculation of Additional Rent based on such measurement for the period prior to the effective date of such adjustment as set out above.

Without limiting the Landlord's rights under this Section 5.3, the Landlord shall be entitled to gross up the Rentable Area of the Premises to take into account the Tenant's right, subject to the terms of this Lease, to use the Amenity Facility (if any) in common with other tenants of the Project as though the Amenity Facility was Common Areas and Facilities (for the purposes of the application of Measurement Standards) for the Project. The Landlord retains the right to modify the size and layout of any such Amenity Facility throughout the Term and any extension or renewal thereof as it sees fit and any change in size of the Amenity Facility will result in the appropriate adjustment as determined by the Landlord to the gross up of the Rentable Area of the Premises.

5.4 Payment of Rent – General

(a) All payments required to be made by the Tenant pursuant to this Lease shall be paid when due, without prior demand and without any abatement, set-off, compensation or deduction whatsoever, except as may be otherwise expressly provided herein, at the address of the Landlord set out in Section 1.1(a)(ii) or at such other place as the Landlord may designate from time to time to the Tenant.

17

- (b) All payments required to be made by the Tenant pursuant to this Lease, except for Rental Taxes, shall be deemed to be Rent and shall be payable and recoverable as Rent, and the Landlord shall have all rights against the Tenant for default in any such payment as in the case of arrears of Rent.
- (c) The Tenant shall pay to the Landlord all Rental Taxes applicable from time to time, calculated and payable in accordance with Applicable Laws and the Tenant shall pay such amount at the earlier of: (i) the time provided for by Applicable Laws; and (ii) the time such Rent is required to be paid under this Lease. The amount payable by the Tenant on account of Rental Taxes shall be deemed not to be Rent for the purpose of such calculation but in the event of a failure by the Tenant to pay any amount, the Landlord shall have the same rights and remedies as it has in the event of a failure by the Tenant to pay Rent.
- (d) At the Landlord's request, the Tenant shall make all payments under this Lease by way of post dated cheques, automatic withdrawals or electronic funds transfer from the Tenant's bank account in Canadian funds and shall execute and deliver either concurrently with this Lease or from time to time within three Business Days following request for it, such documentation as may be required by the Landlord and its bank in order to effect such payments.
- (e) If the Commencement Date is other than the first day of a full period in respect of which any item of Rent is calculated, or the Expiry Date is other than the last day of a full period, then unless otherwise provided in this Lease, the amount of such item of Rent payable in respect of the broken period shall be prorated based on a 365 day year.

5.5 Payment of Additional Rent

(a) Prior to the Commencement Date and at or prior to the beginning of each Fiscal Year thereafter, the Landlord shall compute and deliver to the Tenant a bona fide estimate in respect of such Fiscal Year of the Tenant's share of Property Taxes, the Tenant's Proportionate Share of Operating Costs and such other items of Additional Rent as the Landlord may estimate in advance and the Tenant shall pay to the Landlord in monthly installments one-twelfth of such estimate simultaneously with the Tenant's payments of Basic Rent, provided that the monthly installments on account of the Tenant's share of Property Taxes may be determined so that the Landlord collects all such amounts payable by the Tenant by the final due date in

the relevant calendar year. The Landlord may from time to time re-estimate any items of Additional Rent and may fix monthly instalments for the then remaining balance of the Fiscal Year so that such items will be entirely paid during such Fiscal Year.

- (b) The Landlord shall deliver to the Tenant within a reasonable period of time after the end of each Fiscal Year a written statement or statements (the "Statement") setting out in reasonable detail the amount of Operating Costs, the Property Taxes and such other items of Additional Rent as the Landlord estimated in advance for such Fiscal Year. If the Tenant's share of Property Taxes, the Tenant's Proportionate Share of Operating Costs and other items of Additional Rent actually paid by the Tenant to the Landlord during such Fiscal Year differs from the amount of the Tenant's share of Property Taxes, the Tenant's Proportionate Share of Operating Costs and other items of Additional Rent payable for such Fiscal Year, the Tenant shall pay such difference or the Landlord shall credit the Tenant's account (as the case may be), without interest within 30 days after the date of delivery of the Statement. Failure of the Landlord to render any Statement shall not prejudice the Landlord's right to render such Statement thereafter or with respect to any other Fiscal Year. The Landlord may render amended or corrected Statements.
- (c) The Tenant shall not claim a re-adjustment in respect of Operating Costs or Property Taxes or other items of Additional Rent estimated by the Landlord or the share payable by the Tenant on account thereof for any Fiscal Year except by Notice given to the Landlord within six months after delivery of the Statement, stating the particulars of the error in computation.
- (d) If the Tenant disputes the accuracy of any Statement within the period permitted under Section 5.5(c) above and the Landlord and the Tenant fail to settle the matter within a reasonable period, the matter shall be referred by the Landlord to an Expert for prompt decision. The Tenant shall pay in accordance with the Statement until such decision is rendered. The Expert's signed determination shall be final and binding on both the Landlord and the Tenant. Any adjustment required to any previous payment made by the Tenant or the Landlord by reason of any such determination shall be made within 14 days thereof, and the party required to pay such adjustment shall bear all costs of the Expert, except that if the amount to be paid is 20% or less of the amount in dispute, the Tenant shall pay all such costs.

5.6 Rent Deposit

The Landlord acknowledges receipt from the Tenant of the Rent Deposit to be applied to the Rent as it becomes due or as otherwise provided in Section 1.1(j)(i) and, to the extent it is not so applied from time to time, to be held, without interest, as security (without prejudice to the Landlord's other rights and remedies) for the observance and performance of the Tenant's obligations under this Lease.

5.7 Security Deposit

The Landlord acknowledges receipt from the Tenant of the Security Deposit to be held, without interest, as security (without prejudice to the Landlord's other rights and remedies) for the observance and performance of the Tenant's obligations under this Lease. If the Tenant defaults in the performance of any of the terms, covenants, conditions and provisions of this Lease as and when the same are due to be performed by the Tenant, then the Landlord, at its option, may appropriate and apply all or any part of the Security Deposit on account of any losses or damages sustained by the Landlord as a result of such default. Upon demand by the Landlord following any such appropriation, the Tenant shall pay to the Landlord an amount sufficient to restore the total original amount of the Security Deposit. If the Tenant complies with all of the terms, covenants, conditions and provisions under this Lease and is not then overholding in accordance with Section 16.3, the Security Deposit shall be returned to the Tenant without interest within 60 days after the expiry or earlier termination of the Term, or, at the Landlord's option, shall be applied by the Landlord on account of the last month's Rent.

5.8 Net Lease

The Tenant acknowledges and agrees that it is intended that this Lease shall be a completely carefree net lease for the Landlord and that the Landlord shall not be responsible for any costs, charges, expenses and outlays of any nature whatsoever arising from or relating to the Premises or the BuildingProject during the Term, whether foreseen or unforeseen and whether or not within the contemplation of the parties at the commencement of the Term, except as shall be otherwise expressly provided in this Lease.

ARTICLE 6 - OPERATING COSTS AND TAXES

6.1 Property Taxes Payable by Landlord

The Landlord shall pay all Property Taxes, but it may defer such payments or compliance to the fullest extent permitted by law so long as it pursues in good faith any contest or appeal of any such Property Taxes with reasonable diligence.

6.2 Property Taxes Payable by Tenant

- (a) The Tenant shall pay as Additional Rent directly to the Landlord in each Fiscal Year the Tenant's share of Property Taxes as determined pursuant to this Section 6.2.
- (b) The Tenant's share of Property Taxes shall be the portion of the Property Taxes that are attributable to the Premises, as determined by the Landlord, acting reasonably. Without limiting the foregoing:
 - (i) the Landlord may, if it so elects, determine that the Tenant's share of Property Taxes attributable to the Premises shall be the Proportionate Share of Property Taxes;

- (ii) the Landlord shall be entitled, but not obligated, to allocate Property Taxes amongst categories of premises in the BuildingProject on the basis of such factors as the Landlord determines to be relevant and to adjust the Tenant's share of Property Taxes based on such allocation. Without limiting the generality of the foregoing, Property Taxes may also be allocated in respect of disproportionate contribution to use of parking or consumption of water where the costs of same are included in Property Taxes by the local taxing authority;
- (iii) if there are separate assessments (or, in lieu of separate assessments, calculations made by authorities having jurisdiction from which separate assessments may, in the Landlord's opinion, be readily determined) for the Premises for Property Taxes, the Landlord may have regard thereto;
- (iv) nothing herein shall compel or require the Landlord to adjust, continue to adjust or to make the same determination or allocation of Property Taxes from year to year or in any Fiscal Year; and
- (v) for the purposes of determining the share of Property Taxes payable by the Tenant pursuant to this Lease, Property Taxes shall include such additional amounts as would have formed part of Property Taxes had the BuildingProject been fully assessed during the whole of the relevant Fiscal Year as fully completed and fully occupied by tenants, with no special exemptions or reductions, and without taking into account any actual or potential reduction of Property Taxes or change of assessment category or class for premises within the BuildingProject which are vacant or underutilized.

6.3 Business Taxes and Other Taxes of Tenant

The Tenant shall promptly pay before delinquency to the taxing authorities or to the Landlord, if it so directs, as Additional Rent, any taxes, rates, duties, levies and assessments whatsoever, whether municipal, provincial, federal or otherwise, separately levied, imposed or assessed against or in respect of the operations at, occupancy of, or conduct of business in or from the Premises by the Tenant or any other permitted occupant, including the Tenant's Business Taxes, if levied in the province in which the BuildingProject is situatesituated. Whenever requested by the Landlord, the Tenant shall deliver to the Landlord copies of receipts for payment of all such taxes.

6.4 Assessment Appeals

The Tenant shall not appeal any governmental assessment or determination of the value of the <u>BuildingProject</u> or any portion of the <u>BuildingProject</u> whether or not the assessment or determination affects the amount of Property Taxes or other taxes, rates, duties, levies or assessments to be paid by the Tenant.

6.5 Operating Costs

The Tenant shall pay its Proportionate Share of Operating Costs to the Landlord. Subject to the exclusions and deductions stipulated in Section 6.6, "Operating Costs" means the total, without duplication, of the costs, expenses, fees, rentals, disbursements and outlays (in Sections 6.5 and 6.6 referred to collectively as "costs") of every kind, whether direct or indirect, paid, payable or incurred by or on behalf of the Landlord on an accrual basis (or on a cash basis to the extent that the Landlord determines is reasonable) in the ownership, maintenance, repair, replacement, operation, administration, supervision and management of the BuildingProject, including, without limitation:

- (a) costs of providing security, supervision, traffic control, janitorial, landscaping, window cleaning, waste collection, disposal and recycling and snow removal services and the costs of machinery, supplies, tools, equipment and materials used in connection with the BuildingProject (including rental costs of such items);
- (b) costs of <u>installing and maintaining</u> telecommunications and broadband services and facilities (including riser, rooftop, telephone room <u>and</u>, wireless management, <u>and wifi and mobile networks (such as 5G)</u>), information technology, telecopier, stationery, office equipment, supplies, signs and directory boards and other services and materials required for management, maintenance and operation (whether on or off-site and whether incurred by the Landlord or a management company) <u>of the Project</u>;
- (c) costs of providing electricity, fuel, heat, HVAC, water, telephone, gas, sewage disposal and other utilities and services (including all energy management and administration costs) and costs of replacing Building standard electric light fixtures, ballasts, tubes, starters, lamps, light bulbs, LED lights, and controls (to the extent such item is charged separately to the Tenant pursuant to this Lease then the costs of any such item attributable to other leaseable premises shall be excluded);
- (d) costs of:
 - (i) operating, maintaining, replacing, modifying and repairing the BuildingProject, including without limitation such costs where incurred by the Landlord in order to comply with Applicable Laws or required by the Landlord's insurance carrier or resulting from normal wear and tear to the BuildingProject;
 - (ii) providing, installing, modifying and upgrading energy and water conservation equipment and systems, life safety and emergency response and/or communication systems, materials and procedures and telecommunication and broadband systems and equipment if any;
 - (iii) making <u>Alterations</u> alterations, replacements or additions to the <u>BuildingProject</u> intended to reduce Operating Costs, utility consumption, and/or Greenhouse Gas emissions, improve the operation of the

<u>BuildingProject</u> and the systems, facilities and equipment serving the <u>BuildingProject</u>, or maintain their operation, <u>or improve the Resilience of</u> the <u>Project</u>;

- (iv) replacing machinery or equipment which by its nature requires periodic replacement; and
- (v) developing, and/or modifying and operating the BuildingProject to achieve the objectives of the EnvironmentalSustainability Management Plan, including, without limitation, the costs of data collection (including Landlord Data), reporting, testing, commissioning and re-commissioning the BuildingProject or any part of it;
- (vi) Building Resilience Plan Costs; and
- (vii) implementing and maintaining health and wellness programs, guidelines, policies and procedures, from time to time, in accordance with best practices in the real estate industry in Canada, which may include, without limitation, installing, operating, maintaining, repairing and replacing touchless systems, equipment and facilities, enhanced ventilation systems, equipment and facilities, occupancy monitors, and health screening equipment, as well as implementing health screening procedures and enhanced cleaning and disinfecting of frequently touched surfaces,

all to the extent that such costs are fully chargeable in the Fiscal Year in which they are incurred in accordance with sound accounting principles or practices as applied by the Landlord;

- (e) depreciation or amortization of the costs referred to in Section 6.5(d) above as determined in accordance with sound accounting principles or practices as applied by the Landlord, if such costs have not been charged fully in the Fiscal Year in which they are incurred, and interest on the undepreciated or unamortized balance of such costs, calculated monthly, at an annual rate equal to 5% above the Prime Rate in effect on the first day of each Fiscal Year;
- (f) amounts paid to, or reasonably attributable to the remuneration of, all personnel (whether on or off-site and whether employed by the Landlord or a management company) involved in the maintenance, repair, replacement, operation, administration, supervision and management of the BuildingProject, including fringe benefits, severance pay, termination payments and other employment costs;
- (g) auditing (including compliance audits pursuant to Environmental Laws), accounting, legal and other professional and consulting fees and disbursements incurred in connection with the maintenance, repair, replacement, operation, administration, compliance, supervision and management of the BuildingProject, including those incurred in connection with the EnvironmentalSustainability Management Plan, those incurred with respect to the preparation of the statements

- required under the provisions of this Lease, and those incurred in administering, minimizing, contesting or appealing assessments of Property Taxes (whether or not successful);
- (h) costs of all insurance which the Landlord is obligated or permitted to obtain under this Lease; insurance deductibles, and the amounts of any losses incurred or claims paid either below the insurance deductible amounts or as the co-insurance portion of an insured claim;
- (i) Property Taxes to the extent not charged to the Tenant pursuant to Section 6.2 and to other tenants of the <u>BuildingProject</u> pursuant to lease provisions similar to such Section;
- (j) Capital Tax, Carbon Tax and Carbon Offset Costs;
- (k) fair market rental value (having regard to rent being charged for similar space including additional rent for operating costs and property taxes) of space used by the Landlord and/or its property manager, acting reasonably, in connection with the maintenance, repair, operation, administration and management of the BuildingProject and such fair market rental value of any building amenities (such as conference and day-care facilities provided primarily for tenants of the <a href="Building) Amenity Facility, together with the reasonable costs relating to such building amenities Amenity Facility;
- (I) Electric Car Charging Station Costs attributable to: (i) operating, administering, maintaining, and repairing the Electric Car Charging Stations (including electricity costs if not directly charged to the user of the charging station), provided that the net revenues (if any) received by the Landlord in respect of the operation of such Electric Car Charging Station shall be deducted from such costs; and (ii) installing and replacing the Electric Car Charging Stations, provided that, if the installation or replacement (as the case may be) costs have not been charged fully in the Fiscal Year in which they are incurred, such installation or replacement (as the case may be) costs will be included in Operating Costs on amortized basis in accordance with sound accounting principles or practices as applied by the Landlord, together with interest on the unamortized balance of such installation or replacement as the case may be) costs, calculated monthly, at an annual rate equal to 5% above the Prime Rate in effect on the first day of each Fiscal Year;
- (m) (1)-a management fee in an amount comparable to that which would be charged by a real estate management company for management of similar office buildings in the area in which the BuildingProject is located; and
- (n) the costs of preparing a pandemic risk assessment and/or a Health Emergency Plan, as well as actual costs in dealing with a Health Emergency Costs.

6.6 Limitations on Operating Costs

In determining Operating Costs, the cost (if any) of the following shall be excluded or deducted, as the case may be:

24

- (a) major repairs to structural components that are required as a result of defective design or construction of such structural components;
- (b) interest on, and the capital retirement of debt, except as specifically provided in Section 6.5(e), and ground rent payable to the lessor under any ground or other lease pursuant to which the Landlord has an interest in the BuildingProject;
- (c) other than those expenses incurred in furtherance of the EnvironmentalSustainability Management Plan, expenses relating to decorating or redecorating or renovating rentable space for tenants or occupants of the Building Project and costs relating to tenant inducements, allowances or similar expenses;
- (d) all leasing expenses, real estate brokers' fees, leasing commissions, advertising and space planners' fees;
- (e) repairs or maintenance done for the direct account of other tenants;
- (f) net recoveries by the Landlord in respect of warranties or guarantees and insurance claims to the extent (but only to the extent) that the repair costs in respect of the work covered by such warranties or guarantees or insurance claims have been charged as Operating Costs; and
- (g) amounts recovered from TSPs and tenants as contributions to the cost of telecommunications and broadband related services (including riser, rooftop, telephone room and wireless management) to the extent (but only to the extent) that those costs have been included in Operating Costs.

6.7 Adjustments of Operating Costs

In computing Operating Costs:

- (a) if less than 100% of the Rentable Area of the Building is completed or occupied during any period for which a computation must be made, the amount of Operating Costs will be increased by the amount of the additional costs determined by the Landlord, that would have been incurred had 100% of the Rentable Area of the Building been completed or occupied during that period, provided that the foregoing shall not result in the amount the Tenant pays as its Proportionate Share of such Operating Costs being greater than it would be if the Building was fully occupied and completed;
- (b) where the Landlord determines, acting reasonably but in its sole discretion, that any item(s) of Operating Costs are provided only to or for the benefit of a portion of the

Building, then the Landlord shall be entitled, but not obligated, to allocate the cost of those item(s) over such portion of the Building and adjust the Tenant's Operating Cost payment based on such allocation;

- (c) if the Building is comprised of different categories of leaseable premises, the Landlord shall be entitled, but not obligated, to allocate Operating Costs among the various categories on the basis of such factors as the Landlord determines to be relevant and to adjust the Tenant's Operating Cost payment based on such allocation; and
- (d) if any facilities, services or utilities:
 - (i) for the operation, administration, management, repair and maintenance of the <u>BuildingProject</u> are provided from another building or other buildings owned or operated by Landlord or its manager;
 - (ii) for the operation, administration, management, repair and maintenance of another building or other buildings owned or operated by the Landlord or its manager are provided from the BuildingProject; or
 - (iii) are otherwise shared between the <u>BuildingProject</u> and another building or other buildings,

the costs, charges and expenses of such items shall be allocated by the Landlord, between the **BuildingProject** and other building or buildings on a reasonable basis.

6.8 Reduction or Control of Operating Costs and Utility Consumption

The Tenant shall comply with any practices or procedures that the Landlord, acting reasonably, may from time to time introduce to reduce or control Operating Costs and utility consumption, and shall pay, as Additional Rent, all costs, as determined by the Landlord, that may be incurred by the Landlord as a result of any non-compliance. The Landlord may use an Expert to assist it in making such determination.

ARTICLE 7 - HVAC, UTILITIES AND OTHER LANDLORD SERVICES

7.1 Heating, Ventilating, Air Conditioning, and Utility Consumption

(a) The Landlord shall provide HVAC in quantities and at temperatures required to maintain conditions within a reasonable temperature range in the Premises during Business Hours, for general office uses on an open floor basis and provided that the Density Limit is not exceeded. If Tenant requests HVAC and lighting supplied to for the Premises outside of Normal Business Hours requested by the, Tenant shall be at the Tenant's cost at the rate equal to the then prevailing ratepay for such service or utility to Landlord at then prevailing rate(s) for the Project as determined and separately invoiced by Landlord to Tenant, plus the Landlord's 15% administration fee.

- (b) Any rebalancing of the climate control system necessitated by the installation of partitions, equipment or fixtures by the Tenant or by any use of the Premises not in accordance with the design standards of such system and/or the EnvironmentalSustainability Management Plan shall be performed by the Landlord at the Tenant's expense.
- The Landlord shall not be responsible for inadequate performance of the Building (c) Systems if: (i) attributable to: (A) any arrangement of partitioning in the Premises or changes therein; (B) the failure to shade windows which are exposed to the sun, unless the Building envelope and systems are designed to maintain acceptable environmental conditions without the use of internal window shades; (C) the production by the Tenant of smoke, odours or contaminated air which the Building Systems are not designed to accommodate; (D) any use of electrical power by the Tenant which exceeds the standard of normal use as determined by the Landlord or that as set out in the Environmental Sustainability Management Plan; (E) any use of water by the Tenant which exceeds the normal use as determined by the Landlord or that as set out in the Environmental Sustainability Management Plan (water used to maintain water quality for the Project, such as through flushing programs, shall be included in the normal use amount); (F) any material deterioration in air quality as a result of any furniture, equipment, materials or improvements Alterations located in the Premises or the management practices of the Tenant or any other occupant or Person on the Premises which are not otherwise in accordance with the Environmental Sustainability Management Plan; or (G) supplementary HVAC equipment provided by the Tenant or any other occupant or Person in the Premises; (ii) the occupancy level of the Premises exceeds one Person to every 150 square feet of Rentable Area of the Premises on an open floor basis the Density Limit; or (iii) the Tenant does not keep the HVAC vents or air returns free and clear of all obstructions.

7.2 Electricity and Other Utilities

- (a) The Landlord will provide and permit the Tenant to use the electricity, domestic water, sewage disposal and other utility services serving the BuildingProject in such quantities as the Landlord, from time to time determines to constitute normal use for tenants inof the BuildingProject or as may be specified in the EnvironmentalSustainability Management Plan, and having regard to the Density Limit. The Tenant shall not overload the capacity of any such service. The Tenant shall not bring onto the Premises any Alterations, installations, appliances or business machines which are likely to consume significant amounts of electricity or other utilities or which require special venting without the prior written consent of the Landlord. The Tenant shall not engage any Person to provide any utility service to the Premises.
- (b) The Landlord shall replace Building standard and, at the Landlord's election, non-standard electric light fixtures, ballasts, tubes, starters, lamps, light bulbs and controls in the Premises. In carrying out its obligations, the Landlord may adopt a

system of periodic group re-lamping in accordance with sound building management practices.

- (c) If not separately metered, direct and indirect Tenant will pay all costs relating to the use by the Tenant of electricity and other utility services in quantities which represent normal use for tenants in the Building, as determined by the Landlord, will. Such costs may form part of Operating Costs or be paid by the Tenant to, if the Landlord separately as Additional Rent, as and to the extent that the Landlord may electelects from time to time, may be charged to the Tenant on a separate basis, together with an administration fee equal to 15% of such costs. The Landlord may install, at the Tenant²'s expense, separate and multiple water, electricity and natural gas (if applicable) meters or other measuring devices throughout the Premises or elsewhere to measure the Tenant²'s consumption of each such utility utilities and the Landlord may use an Expert to assist it in determining such consumption. Any costs for third party monitoring services shall be included in Operating Costs.
- (d) The Landlord shall replace any Building standard ballasts, tubes, starters, lamps, light bulbs, LED lights and controls in the Premises at the Tenant's cost, payable as Additional Rent, unless such costs are included in Operating Costs. In carrying out its obligations, the Landlord may adopt a system of periodic group re-lamping in accordance with sound building management practices.
- (e) If electricity, propane, natural gas or water is separately billed to the Premises or the Tenant, then the Tenant shall provide to the Landlord from time to time copies of all such electricity, propane, natural gas or water bills within 10ten Business Days of receipt. Upon the Landlord's request, the Tenant shall deliver to the Landlord an authorization from the Tenant addressed to each utility supplier permitting such supplier to share the Tenant's consumption data with the Landlord directly.

7.3 Special HVAC Services and Utilities and Excess Quantities

(a) The Landlord shall have no obligation to provide the Tenant with HVAC, electricity, sewage disposal, water or other utility services of a type or in quantities that exceed normal use by tenants inof the BuildingProject (as such normal use is determined by the Landlord, and having regard to the Density Limit), or that exceed those limits set out in the EnvironmentalSustainability Management Plan, unless the Landlord determines, in its sole discretion, that the provision of such services:

(ai) is within the capacity of the Building Systems; (bii) would not affect the operation, aesthetics or structure of the Building or any other part of the Project; (eiii) would not reduce the efficiency of the existing services supplied to other tenants or parts of the BuildingProject; (div) is otherwise feasible; (v) would not be subject to a tax, fee, levy or other charge from an authority having jurisdiction; and (evi) is consistent with the EnvironmentalSustainability Management Plan and the Building Resilience Plan. The Tenant will pay to the Landlord all costs, both non-recurring and recurring, of providing all such services. Such costs shall be determined by the Landlord in a reasonable manner, which may include installation

28

- at the Tenant's expense of separate and multiple water, electricity and natural gas (if applicable) meters or other measuring devices in the Premises or elsewhere or the Landlord may use an Expert to assist it in determining such costs.
- The Landlord will not be required to furnish electrical current for equipment that (b) requires more than 240 volts or other equipment whose electrical energy consumption exceeds normal office usage (and having regard to the Density Limit). If the Tenant's requirements for or consumption of electricity exceed the electricity to be provided by the Landlord as described herein, the Landlord will, at the Tenant's expense, make reasonable efforts to supply such service through thenexisting feeders and risers serving the Project and the Premises if, in the Landlord's judgment, the same are necessary and will not cause permanent damage to the Project, Building or Premises, cause or create a dangerous or hazardous condition, entail excessive or unreasonable alterations, repairs or expenses, adversely affect the Landlord's ability to provide reasonable service to the balance of the Project, or unreasonably interfere with or disturb other tenants of the Project. If the Tenant uses machines or equipment in the Premises that abnormally affect the temperature otherwise maintained by the air conditioning system or otherwise overload any utility, the Landlord may, or may require the Tenant to, install supplemental air conditioning units or other supplemental equipment in the Premises or, if necessary, elsewhere in the Project, and the cost thereof, including the cost of design, installation, operation, use and maintenance, repair and replacement, in each case plus an administrative fee of 15% of such cost, will be paid by the Tenant, as Additional Rent.

7.4 Other Landlord Services

- The Landlord shall provide janitorial services to the Premises in accordance with (a) standards from time to time prevailing for similar office buildings in the area in which the **Building**Project is located, subject to the **Environmental**Sustainability Management Plan. Notwithstanding the foregoing, the Landlord may choose to provide such cleaning services during normal Business Hours to assist in achieving the objectives of the Environmental Sustainability Management Plan, provided that if the Landlord does so choose, it shall use its reasonable commercial efforts to minimize the disruption of such cleaning services to the business and operations of the Tenant. The Tenant shall grant access necessary for the performance of the janitorial services and shall leave the Premises in a condition that facilitates the performance of such services. Other than as included in janitorial services, all curtains, carpets, rugs and drapes of any kind in the Premises shall be cleaned and maintained by the Tenant in accordance with the Environmental Sustainability Management Plan. The Tenant shall not engage any Person to provide cleaning or janitorial services to the Premises without the Landlord's written consent.
- (b) The Landlord shall provide elevator service during Business Hours and additional hours for use by the Tenant in common with others, except when prevented by maintenance or repairs. Subject to emergencies, the Landlord will operate at least one passenger elevator for use by tenants at all times.

(c) The Landlord shall provide necessary supplies in public washrooms sufficient for normal use by tenants <u>inof</u> the <u>BuildingProject</u>.

7.5 Additional Services Provided by Landlord

Wherever this Lease provides that the Tenant is to pay a cost or expense to the Landlord as an item of Additional Rent (except for Operating Costs and the 15% administration fee referred to herein), the Tenant shall pay, in addition to such cost or expense, the Landlord's administration charge of 15% of such cost or expense, which cost shall also be an item of Additional Rent. The Tenant shall pay to the Landlord the costs of all such services provided at the Tenant's request or otherwise provided for herein and which are not included in Operating Costs including, without limitation: (a) the provision of HVAC and lighting, electricity, water, steam and/or chilled water and other utilities and services outside of Business Hours or of a special nature or in excess quantities in accordance with Section 7.3 of this Lease; (b) replacement of non-standard electric light fixtures, ballasts, tubes, starters, lamps, light bulbs, LED lights, and controls; (c) special janitorial or cleaning services; (d) operating freight elevators used or reserved for the sole benefit of the Tenant outside Normal Business Hours and supervising the movement of furniture, equipment, freight and supplies for the Tenant; and (e) construction of any Leasehold Improvements or other work performed at the request of or on behalf of the Tenant; (f) costs for Building services, maintenance, repairs and/or replacements which are necessitated by reason of the particular use or occupancy of the Premises or any of the Common Areas and Facilities by the Tenant, its licensees, corporate or private members, employees, agents or persons having business with the Tenant at the Premises; and (g) other services performed at the request of or on behalf of the Tenant.

7.6 Telecommunications

- (a) The Landlord shall incur no expense or liability whatsoever with respect to any aspect of the provision of telecommunication services, including, without limitation, the cost of installation, service, materials, repairs, maintenance, interruption or loss of telecommunication service.
- (b) The Tenant may utilize a telecommunication service provider (a "TSP") of its choice with the Landlord's prior written consent, but:
- (c) if the TSP is required to provide or install facilities in the Building in order to enable it to provide service to the Tenant, the Landlord must first determine that so long as there is sufficient space in, or on the Building for the installation of the TSP's facilities and that the TSP is acceptable to the Landlord;
- (d) <u>(ii)</u> <u>if the TSP intends to install, or has installed or purchased facilities situated in the Building for the purpose of providing telecommunication services to tenants in the Building, the. The Landlord may require the TSP to execute and deliver the Landlord's standard form of TSP licence agreement;</u>

- (iii) the Tenant shall be responsible for all costs incurred by the Landlord in enabling usage by the Tenant of its choice of TSP not otherwise paid by such TSP; and
- (e) (iv) the Tenant shall be responsible for the removal of all wiring serving the Premises by such TSP at the expiry of the Term, together with any cable wire, if required by the Landlord.(c) The Landlord may deem it desirable to provide a central telecommunications cable distribution system ("CDS") in the Building for use by TSPs and tenants. If the Landlord provides a CDS, the Tenant's TSP or the Tenant, as the case may be, may be required by the Landlord to use the CDS for its communications cabling needs on terms and conditions to be set by the Landlord. These terms and conditions will include obligations for the TSP, or the Tenant, as the case may be, to pay costs and to contribute to Operating Costs associated with the CDS and a complete release of the Landlord and indemnity from the TSP or the Tenant, as the case may be, in respect of the use of the CDS.
- (f) (d) If the Tenant's approved TSP does not have a point of connection in the Premises, the Tenant may be required to install its own cable and facilities or to purchase cable and facilities from the Landlord for installation in the communication pathways and/or risers of the Building for connection to the Tenant's TSP's facilities in the main terminal room, at the main distribution frame or at other points of connection designated by the Landlord. In such case: (i) the The Tenant may be required to pay all costs incurred by the Landlord; (ii) the Tenant may be required to remove such cable and facilities and restore any damage caused by the removal at the Landlord's option, or to pay the cost of removal and restoration at the end of the service term of that TSP; (iii) the Tenant may be required to contribute to the costs of riser management incurred by the Landlord; and (iv) the Tenant may be required to and its TSP shall abide by any policies, directions or requirements of any riser manager retained by the Landlord and to pay, in addition, any direct costs invoiced/or its riser manager (if any). The Tenant shall be responsible for any costs incurred by the Landlord to enable the TSP to provide or discontinue services to the Tenant by such (including costs of riser manager in respect of management, installing and maintaining wifi services and 5G services, plan review charges, inspection charges of facilities, and other services provided by such riser manager toof installing or removing cables and wiring). Upon 30 days' prior Notice, the Landlord may require that all or any portion of the cables or facilities installed by the Tenant or its TSP be relocated at the Tenant's cost.
- If the Landlord decides to provide a central telecommunications cable distribution system ("CDS") for use by TSPs and tenants, which may be wired or wireless, the Landlord may require the Tenant or its TSP to use the CDS on terms and conditions set by the Landlord, including payment of costs associated with the CDS and a full release and indemnity of the Landlord from the TSP and the Tenant for the use and operation of the CDS.

- (h) (e) If required by the The Landlord, the Tenant shall change its TSP if the licence agreement referred to above in Section 7.6(b) is terminated or expires and is not renewed is not liable for any aspect of providing telecommunication services, including the cost of installation, service, materials, repairs, maintenance, interruption or loss of telecommunication service. The Tenant acknowledges that the Landlord has no obligation to ensure continuation of services by the Tenant's TSP or any other TSP in the Building.
- (f) The Landlord may require, upon 30 days prior written Notice, that the Tenant relocate all or any portion of the cables or facilities installed by it or its TSP.

7.7 Signs and Premises Identification

The Tenant shall not erect, affix, install or maintain any signs, lettering, identification or any promotional or other written materials visible from the exterior of the Building or from any interior Common Areas and Facilities. The Landlord shall, at the request and expense of the Tenant, supply and install: (a) on or near the entrance door of the Premises a sign bearing the name of the Tenant; (b) identification in any elevator lobby directional signage on the Tenant's floor; and (c) one entry in any directory board for the Building, each in accordance with the Landlord's uniform scheme for identification signage. Any tenant occupying at least a full floor in the Building may, subject to having received the Landlord's prior written approval as to design, location, material and method of installation, supply and install its own sign in the elevator lobby of each full floor occupied by it. All such signage shall comply with the applicable provisions of the Environmental Sustainability Management Plan, including, without limitation, those pertaining to light pollution reduction, and migrating bird safety programs.

ARTICLE 8 - OPERATION, CONTROL AND MAINTENANCE BY LANDLORD

8.1 Operation of the **Building**Project by Landlord

The Landlord shall operate the <u>BuildingProject</u> in accordance with all Applicable Laws and with standards from time to time prevailing for similar office buildings in the area in which the <u>BuildingProject</u> is located, subject, however, to the limitations occasioned by the design and age of the <u>BuildingProject</u> and the capacity of the Building Systems, and the provisions of this Lease.

The Landlord shall also manage, maintain, operate, repair, and upgrade the <u>BuildingProject</u> as necessary in accordance with the <u>EnvironmentalSustainability</u> Management Plan.

The Landlord's costs of compliance with this Section 8.1 shall be included in Operating Costs, to the extent provided therein.

8.2 Control of the **BuildingProject** by Landlord

The Landlord has at all times exclusive control of the <u>BuildingProject</u> and its management and operation, but not so as to deny the Tenant access to the Premises except in an emergency. Without limiting the generality of the foregoing, at any time and from time to time, the Landlord may:

- (a) make repairs, replacements, changes or additions to the structure, systems, facilities and equipment in <u>or on</u> the <u>BuildingProject</u> (including the Premises) where necessary to serve the Premises or other parts of the <u>BuildingProject</u> or to achieve the objectives of the <u>EnvironmentalSustainability</u> Management Plan;
- (b) make changes or additions to any part of the <u>BuildingProject</u> not in or forming part of the Premises including, without limitation, dedicating or conveying portions of the Lands, granting easements, rights-of-way, restrictive covenants or other interests in the Lands and constructing additional improvements in or adjoining the Lands;
- (c) terminate, <u>limit</u>, <u>restrict</u> or amend the Tenant's right of use of any of the Common Areas and Facilities <u>(including, without limitation, in connection with a Health Emergency)</u>, change the location and size of any of the Common Areas and Facilities or use parts of the Common Areas and Facilities for promotional or other activities:
- (d) retain contractors and employ all personnel, including supervisory personnel and managers, that the Landlord considers necessary for the effective maintenance, repair, operation, management and control of the BuildingProject;
- (e) <u>collect and maintain ownership of the Landlord Data and the Landlord may alter</u> the operation, management, maintenance, and repair of the Project, acting reasonably, as a result of such Landlord Data; and
- (f) (e) do and perform such other acts in and to the <u>BuildingProject</u> or any of its component parts as the Landlord considers reasonable for the proper and efficient maintenance, repair, operation, management and control of the <u>BuildingProject</u>,

provided that in the course of the Landlord's exercise of its rights hereunder, the Landlord shall be deemed not to have re-entered the Premises nor to have breached any obligation of this Lease. The Landlord shall perform all of its work as expeditiously as is reasonably possible so as to interfere as little as is reasonably possible with the Tenant's use of the Premises.

8.3 Name of **BuildingProject**

The Landlord may from time to time designate a name or other identification for the Building and/or Project. The Tenant shall be responsible for any costs it incurs as a result of any changes in the name or identification (such as changes to its stationery and other material). The Tenant shall have no rights in any such names or identification.

8.4 Maintenance and Repair by Landlord

The Landlord shall keep or cause to be kept the following in good repair to the standards from time to time prevailing for similar office buildings in the area in which the <u>BuildingProject</u> is located subject, however, to the limitations occasioned by the design and age of the Building and the capacity of the Building Systems and to reasonable wear and tear not inconsistent with such standard (the "Repair Standard"):

- (a) the footings, foundations, structural columns and beams, structural subfloors, bearing walls, exterior walls, windows and roofs of the Building; and
- (b) the Common Areas and Facilities, (the "Repair Standard") including in accordance with Schedule "E" of this Lease),

provided that:

- (c) if all or part of Building Systems require repair, replacement, maintenance or inspections in order to comply with the Repair Standard, the Landlord shall have a reasonable time in which to complete such work, and during such time shall only be required to maintain such services as are reasonably possible in the circumstances;
- (d) no reduction or discontinuance of such services or loss of use of the Premises shall be construed as an eviction of the Tenant or (except as specifically provided in this Lease) release the Tenant from any obligation under this Lease; and
- (e) the Repair Standard may be suspended from time to time due to a Health Emergency.

8.5 Access by Landlord

The Tenant shall permit the Landlord, its agents and others authorized by it, to enter the Premises (and to have access to any Building Systems) to inspect, to provide services or to make repairs, replacements, changes or Alterations as set out in this Lease, to take such steps as the Landlord may deem necessary for the safety, improvement, alteration or preservation of the Premises or the BuildingProject or to take such steps as may be reasonably necessary to comply with the EnvironmentalSustainability Management Plan and to show the Premises to Mortgagees, prospective Mortgagees, purchasers and prospective purchasers and, during the last 18 months of the Term, to prospective tenants. In carrying out such rights the Landlord shall use reasonable efforts to minimize interference with the Tenant's use and enjoyment of the Premises. The Landlord shall whenever possible give reasonable Notice to the Tenant prior to such entry (other than in the case of an emergency or apprehended emergency), but no such entry shall constitute a re-entry by the Landlord or an eviction or entitle the Tenant to any abatement of Rent.

8.6 Relocation

The Landlord shall have the right from time to time, on not less than 60 days. Notice to the Tenant, to relocate the Premises to other premises within the Building having approximately the same area as the Premises. If the Landlord relocates the Premises prior to occupancy by the Tenant, it shall reimburse the Tenant for all expenses already incurred by the Tenant in preparing to move into the Premises to the extent that such expenditure is for items or materials not usable in the alternate premises. If the Landlord relocates the Tenant after occupancy by the Tenant, the Landlord shall provide the relocated premises improved to a standard and using materials of approximately the same quality as the Leasehold Improvements which exist in the existing Premises at the time of relocation and reimburse the Tenant (upon receipt of copies of receipted third party invoices) for

direct costs associated with the relocation, including, without limitation, moving costs, reprinting of a limited supply of stationery and supplies and disconnection and reconnection of telephone and computer equipment and systems. In no case will the Tenant be reimbursed or compensated for indirect costs including overhead, overtime charges or loss of profits and the Tenant will minimize costs by re-using all fixtures, equipment and trade fixtures from the Premises where it is feasible to do so. The Landlord agrees to use reasonable efforts to effect the relocation with a minimum of disruption to the Tenant's business. The Landlord and the Tenant shall enter into a lease amending agreement in the Landlord's standard form to confirm the terms of the relocation including, without limitation, any adjustment to the Basic Rent if the Rentable Area of the relocated premises is different than the Rentable Area of the existing Premises and to confirm that all other terms and conditions of this Lease shall apply with respect to the relocated premises for the remainder of the Term.

8.7 Health Emergency

If a Health Emergency exists, the Landlord may, without liability or compensation to the Tenant, amend, supplement or otherwise enforce any existing Health Emergency rules or regulations in existence, may impose additional rulesRules and Regulations (including in respect of mandatory personal protective equipment, physical distancing regulations and temperature checks), and may impose restrictions to mitigate or minimize the effects of the Health Emergency. Without limiting the generality of the foregoing:

- during a Health Emergency, the Landlord shall be entitled to restrict or limit access to the Building to employees of the Tenant only, and/or to prohibit entry by visitors or invitees for a reasonable period of time during such event. In addition, the Landlord shall be permitted to restrict or limit access to or use of the Building, the Common Areas and Facilities or Building Systems, and the Landlord may implement safety procedures in connection with access to the Building including, by way of example only, health screening procedures and queuing procedures (including, without limitation, staggering the entry and exit times of tenants and their employees);
- (b) during a Health Emergency, the Landlord shall also be entitled to specify specific modes of ingress and egress from and to the Building for tenants generally, or for specific tenants, occupants or invitees who may have a heightened risk of either exposure to a health threat or a heightened risk of transfer of unhealthy condition to other tenants, invitees or visitors in the Building;
- (c) the Landlord shall be entitled during a Health Emergency to close all or any part of the Building if it determines that it is not safe to continue to operate the Building or certain parts of the Building (including, without limitation, limiting access to elevators and public washrooms);
- (d) the Landlord shall be entitled to place signage and labelling in and around the Premises, as required, to direct the movement of individuals in order to comply with public health guidelines (such as, by way of example only, designating which

- doors to use to enter and exit a space, and arrows to indicate flow of occupants through the space);
- (e) the Landlord shall be entitled, during such time as there is a Health Emergency or a Health Emergency Plan is in effect, to require all occupants to comply with reasonable measures imposed in respect thereof by the Landlord, including health screening, the use of hand washing and other sanitation products directly related to the management of the health threat, attendance at mandatory training sessions, and the use of additional protective clothing by all occupants, invitees and tenants such as protective barriers, gloves and masks;
- (f) the Tenant shall participate in any Health Emergency drill that the Landlord shall choose to implement acting reasonably, in preparation for a Health Emergency; and
- (g) the Landlord shall have the right during a Health Emergency to require the Tenant to decontaminate all or any part of the Premises, failing which the Landlord shall be entitled to enter the Premises and do so at the Tenant's expense. Any steps that the Landlord may choose to take are in its sole and unfettered discretion and nothing herein shall obligate the Landlord to effect any such decontamination; and.
- (e) the Landlord shall be entitled during a Health Emergency to close all or any part of the Building if it determines that it is not safe to continue to operate the Building or certain parts of the Building

Notwithstanding anything to the contrary: (i) the Tenant shall, immediately upon becoming aware of same, notify the Landlord and public health authorities of any outbreak of an infectious disease amongst its employees where such outbreak may impact the health and/or safety of other tenants of the Project or lead to a Health Emergency; and (ii) in the event of any Health Emergency, the Tenant shall, immediately upon becoming aware of same, notify the Landlord and public health authorities if any of its employees, contractors, invitees or other persons who have visited the Premises during such Health Emergency become infected by the applicable disease, virus or other biological or physical agent that is the subject matter of the Health Emergency.

8.8 Amenity Facility

(a) The Landlord hereby grants to the Tenant the non-exclusive license to enter into and use any Amenity Facility in common with other tenants of the Project during the Term and any extension or renewal thereof, at the sole risk and liability of the Tenant subject to any Rules and Regulations and other terms and conditions imposed by the Landlord (including, without limitation, the execution of any waiver(s) requested by the Landlord). The term of this license shall be coincidental with the Term, as the same may be extended or renewed, the intent being that this license will terminate on the date this Lease expires or is otherwise terminated, provided that any such Amenity Facility may be modified and/or eliminated from time to time, and such modification and/or elimination shall not constitute a breach of this Lease, nor shall the Landlord be liable to the Tenant for any such modification and/or elimination.

- (b) Without restricting the rights and remedies of the Landlord as otherwise set out in this Lease, should any employee (or employees) of the Tenant fail to comply with any Rules and Regulations regarding the access to or use and operation of the Amenity Facility and should such failure:
 - (i) continue for a period of five days after Notice to the Tenant from the Landlord; or
 - (ii) occur on more than three occasions,

the Landlord may without further Notice to the Tenant terminate the license referred to in Section 8.8(a) and all rights of the offending employee (or employees) to use any such Amenity Facility under this Section 8.8 shall cease.

ARTICLE 9 - MAINTENANCE AND ALTERATIONS BY TENANT

9.1 Maintenance and Repair by Tenant

The Tenant shall at its sole cost manage, maintain, operate and repair the Premises and all Leasehold Improvements in good order and condition to the standards from time to time prevailing for similar office buildings in the area in which the BuildingProject is located subject to reasonable wear and tear not inconsistent with such standard and with the exception only of those repairs which are the obligation of the Landlord under this Lease, and subject to Article 14. The Tenant shall also manage, maintain, operate and repair the Premises and construct, use, operate and maintain Leasehold Improvements and all furnishings, fixtures and equipment located in the Premises so as to comply with the EnvironmentalSustainability Management Plan. Where there is a conflict between the provisions of the EnvironmentalSustainability Management Plan, and those standards from time to time prevailing for similar office buildings in the area in which the BuildingProject is located, the provisions as set out in the EnvironmentalSustainability Management Plan shall prevail. <a href="[Optional clause: Tenant agrees, at its sole cost and expense, to obtain and maintain throughout the Term LEED Commercial Interiors ("CI") designation for the Premises, and all repairs, maintenance and Alterations undertaken by the Tenant shall be done so in a manner consistent with the maintenance of such designation]

9.2 Alterations by Tenant

The Tenant may from time to time at its own expense install Leasehold Improvements and alter existing Leasehold Improvements consistent with the Environmental Sustainability Management Plan (the "Alterations") provided that:

(a) all Alterations shall require the prior written approval of the Landlord, save and except for minor Alterations to Leasehold Improvements which do not affect the structure of the Building, any exterior walls, windows or roof, any of the Building Systems, the EnvironmentalSustainability Management Plan, or the aesthetics of the Building and which do not require a building permit provided the Tenant has given Notice with reasonable detail of the proposed Alterations to the Landlord in advance;

- (b) for Alterations which require the Landlord's approval, the Tenant shall furnish the Landlord with two complete sets digital copies (in the format requested by the Landlord) of professionally prepared working drawings (which shall include any architectural, structural, electrical, mechanical, computer system wiring and telecommunication plans) of the proposed Alterations and a statement as to how such Alterations are consistent with the Environmental Sustainability Management Plan and the Tenant Construction Manual. The Landlord may require the Tenant shall to retain the Landlord's base building mechanical, electrical, environmental and structural engineering consultants to ensure compatibility of the Alterations with the Building Systems, the Environmental Sustainability Management Plan and the Tenant Construction Manual. If the Tenant uses other consultants for the preparation of the Tenant's working drawings, then the Landlord may elect to retain architects, environmental consultants and engineers to review such working drawings for the purpose of approving the proposed Alterations (it being understood that notwithstanding such approval, the Landlord shall have no responsibility with respect to the adequacy of such working drawings). The Tenant shall pay to the Landlord, on demand, the costs of the examination of such drawings by either the Landlord or an outside consultant plus an administration fee of 15% of such costs:
- (c) the Alterations shall be subject to the reasonable regulations, supervision, control and inspection by the Landlord and, in addition to any other payment contained in this Article, the Tenant shall pay to the Landlord, on demand, the Landlord's then current fee for coordination services provided by the Landlord during the Tenant's construction of its Alterations:
- (d) the Tenant shall provide, prior to the commencement of Alterations, evidence of required workers' compensation coverage and proof of owner and contractors protective liability insurance coverage, with the Landlord, any property manager and any Mortgagee as required by the Landlord, to be named as additional insureds, in amounts, with insurers, and in a form reasonably satisfactory to the Landlord, which shall remain in effect during the entire period in which the Alterations will be carried out. In addition, if reasonably requested by the Landlord, the Tenant shall provide proof of performance and payment bonds being in place;
- (e) the Tenant will deliver a list identifying every contractor and subcontractor, accompanied by an up-to-date valid clearance certificate for each of them issued by the appropriate workers' compensation, safety and insurance authority and the Landlord shall have approved, prior to commencement of the Alterations, such contractors and subcontractors and their respective labour affiliations. The Tenant will not use any contractor or permit the use of any sub-contractor that is not identified on the list;
- (f) if any proposed Alterations could affect the structure <u>or</u> the exterior walls <u>of the</u> <u>Building</u> or the Building Systems, the Landlord may require that any such

- Alterations be performed by either the Landlord or its contractors in which case the Tenant shall pay the Landlord's cost plus an administration fee of 15%;
- (g) the Tenant shall have provided to the Landlord a copy of the contract for the Alterations and evidence satisfactory to the Landlord as to the existence of all necessary permits;
- (h) the Tenant shall perform the Alterations or cause the Alterations to be performed:

 (i) in accordance with any construction methods and procedures manual for the
 BuildingProject including the Tenant Construction Manual and the
 EnvironmentalSustainability Management Plan; (ii) in accordance with the plans
 and specifications submitted to and approved by the Landlord; (iii) in accordance
 with any conditions, regulations, procedures or rules imposed by the Landlord; (iv)
 in compliance with all Applicable Laws; and (v) in a good and workmanlike and
 expeditious manner;
- (i) the Tenant shall ensure that all cabling installed in the Building in connection with Tenant's business in or use of the Premises is appropriately labeled. For greater certainty, installation of flammable cabling shall be strictly prohibited;
- (j) the Landlord may inspect construction as it proceeds;
- (k) upon completion of the Alterations, the Tenant shall provide the Landlord with a complete set of "as built" drawings for the Alterations; and
- (l) if the Tenant fails to observe any of the requirements of this Article, the Landlord may require that construction stop and, at the Landlord's option, that the Premises be restored to their prior condition failing which the Landlord may do so and the Tenant shall pay the Landlord's cost plus an administration fee of 15%.

9.3 Removal of Improvements and Fixtures

All Leasehold Improvements shall immediately upon their placement become the Landlord's property without compensation to the Tenant. Except as otherwise agreed by the Landlord in writing, no Leasehold Improvements or trade fixtures shall be removed from the Premises by the Tenant either during or at the expiry or earlier termination of the Term except that:

- (a) the Tenant may, during the Term, in the usual course of its business, remove its trade fixtures, provided that the Tenant is not in default under this Lease; and
- (b) the Tenant shall, at its sole cost if required by the Landlord: (i) remove all of its trade fixtures; and (ii) remove such of the Leasehold Improvements and wiring, cables and related devices and equipment in the Premises and restore the Premises to the then current base building standard of the Building, all as the Landlord shall require by Notice prior to the expiration of the Term. Such removal and restoration shall be completed by the later of: (A) the end of the Term; and (B) 15 days after the Landlord's Notice, provided that in the event of termination of this Lease prior

to the expiry of the Term, such removal and restoration shall be completed no later than 15 days after the date the Landlord recovers possession of the Premises.

Any removal by the Tenant of any trade fixtures, personal property and/or Leasehold Improvements as permitted by this Section 9.3 shall be completed by or on behalf of the Tenant in accordance with the EnvironmentalSustainability Management Plan and the Tenant Construction Manual.

The Tenant shall at its own expense repair any damage caused to the BuildingProject by the Leasehold Improvements, trade fixtures or wiring, cables and related devices and equipment and/or such removal and restoration. If the Tenant does not remove its trade fixtures, or wiring, cables and related equipment prior to the expiry or earlier termination of the Term, such trade fixtures or wiring, cables and related devices and equipment shall, at the option of the Landlord, be deemed abandoned and become the property of the Landlord and may be removed from the Premises and sold or disposed of by the Landlord in such manner as it deems advisable and the Tenant shall pay to the Landlord on demand all costs incurred by the Landlord in connection therewith, plus an administration fee of 15% of the costs. If the Tenant fails to complete any work referred to in this Section within the period specified, the Tenant shall pay compensation to the Landlord for damages suffered by the Landlord for loss of use of the Premises, which damages shall not be less than 150% of the per diem Rent payable during the last month preceding the expiry or earlier termination of the Term.

9.4 Liens

The Tenant shall pay before delinquency for all materials supplied and work done in respect of the Premises so as to ensure that no lien or claim of lien is registered against any portion of the LandsProject in respect of materials supplied to or work done for the Tenant in respect of the Premises or against the Landlord's or Tenant's interest in the LandsProject. If a lien or claim of lien is registered or filed, the Tenant shall discharge it at its expense within five Business Days after Notice from the Landlord (or sooner if such lien or claim is delaying a financing or sale of all or any part of the LandsProject), failing which the Landlord may at its option discharge the lien or claim of lien by paying the amount claimed to be due into court and the amount so paid and all expenses of the Landlord including legal fees (on a solicitor and client basis) shall be paid by the Tenant to the Landlord. The Tenant shall not mortgage, charge, grant a security interest in or otherwise encumber any Leasehold Improvements.

9.5 Notice by Tenant

The Tenant shall promptly notify the Landlord of any accident, defect, damage or deficiency which occurs or exists in any part of the Premises, the Building Systems within the Premises or the Common Areas and Facilities located on the floor(s) on which the Premises is located and which comes to the attention of the Tenant.

ARTICLE 10 - USE OF PREMISES

10.1 Permitted Use

The Tenant shall continuously use the whole of the Premises only as a business office, which the Tenant shall operate in a first-class, reputable manner befitting the reputation and image of the BuildingProject, and for no other purpose. The Tenant shall not use the Premises in a manner which does or could result in excessive demands being placed on the Building Systems or other Common Areas and Facilities, and shall use the Premises in a manner consistent with the EnvironmentalSustainability Management Plan. Without limiting the generality of the foregoing, the Tenant shall not use the Premises in a manner that might imperil any existing or targeted certification or accreditation of the BuildingProject (or any part thereof) as more particularly described in Schedule "E", Section 1.2 or any specific objective listed in Section 1.3 of Schedule "E" herein of this Lease (including Sections 1.2 and 1.3 thereof).

10.2 Compliance with Laws

The Tenant shall use and occupy and shall cause the Premises to be used and occupied in compliance with all Applicable Laws and in a safe, careful and proper manner. It is the Tenant's responsibility to ensure that its use from time to time is permitted by all Applicable Laws. At the Landlord's request the Tenant shall comply with any directive, policy or request of any governmental or quasi-governmental authority or any other reasonable request of the Landlord, in respect of any energy conservation, water conservation, waste management, health, safety, security or other matter relating to the operation of the BuildingProject. If due primarily to the Tenant's use or occupancy of the Premises, improvements or changes are necessary to comply with any Applicable Laws or with any such directive, policy or request or with the requirements of insurance carriers, the Landlord may at its option either do the necessary work, at the expense of the Tenant, or forthwith give Notice to the Tenant to do such work within the requisite period of time and the Tenant shall then do such work within the requisite period of time. The Tenant shall pay to the Landlord the costs of any such work done by the Landlord.

10.3 Nuisance, Interference, Waste, Overloading

The Tenant shall not bring into the Premises any Hazardous Substance or other contaminant without the written consent of the Landlord, which may be arbitrarily withheld. Notwithstanding the foregoing to the contrary, Tenant shall be entitled to use Hazardous Substances in quantities, and for purposes, that are typical in general business office use, provided that same are used, handled, stored, transported and disposed of in strict compliance with all Environmental Laws.

The Tenant shall not cause or allow any act or thing which constitutes a nuisance or which is offensive to or which constitutes a health hazard to the Landlord or other occupants of the BuildingProject or which interferes with the operation of any Building Systems or with the computer equipment, telecommunication equipment or other technological equipment of the Landlord, any service providers or other occupants of the BuildingProject. The Tenant shall keep the Premises free of debris and other items that might attract rodents or vermin and free of anything of a dangerous, noxious or offensive nature or which could create a fire, environmental, health or other hazard (including any electromagnetic fields or other forms of radiation) or undue vibration,

heat or noise. The Tenant shall not cause or allow any overloading of the floors of the Building or the bringing into any part of the Building Project, including the Premises, of any articles or fixtures that by reason of their weight, use, energy consumption, water consumption or size might damage or endanger the structure of the Building or any of the Building Systems. The Tenant shall take reasonable steps to ensure (a) minimal transmission of noise or emission of smoke, light and odour from the Premises to other premises within the Building Project; and (b) minimal transmission of electromagnetic radiation from the Premises in respect of any particular equipment located within the Premises or elsewhere within, upon or beside the Building Project but used solely by the Tenant including, without limitation, any rooftop antenna(e).

10.4 Rules and Regulations

The Tenant shall comply and cause every Person over whom it has control to comply with the Rules and Regulations and the EnvironmentalSustainability Management Plan. The Landlord shall have the right from time to time to make reasonable amendments, deletions and additions to such Rules and Regulations. If the Rules and Regulations conflict with any other provisions of this Lease, the other provisions of this Lease shall govern. The Landlord shall not be obligated to enforce the Rules and Regulations and shall not be responsible to the Tenant for failure of any Person to comply with the Rules and Regulations or any environmental management planSustainability Management Plan applicable to the BuildingProject. The Rules and Regulations may differentiate between different types of tenants, different parts of the BuildingProject or otherwise. The Landlord agrees that it will not enforce the Rules and Regulations in a manner that is discriminatory to the Tenant.

10.5 Density Limit

In connection with its use and occupation of the Premises, the Tenant shall not, at any one time, exceed a population density limit of one (1) individual per square feet of Rentable Area of the Premises on an open floor basis (and if the Premises is comprised of more than one (1) floor, the Tenant shall not exceed a population density limit of one (1) individual per square feet of Rentable Area of the Premises on each floor) (the "Density Limit"). Despite the foregoing, the Density Limit may, from time to time, exceed such number on a temporary basis for occasional meetings of a temporary nature, provided that the Tenant shall comply with Applicable Laws in respect of population density and shall be responsible for addressing any excess utility, HVAC and other service or Building System requirements as set forth in this Lease and the Tenant shall be responsible for all costs in connection therewith. The Tenant shall also be responsible for paying all third party costs incurred by the Landlord as a result of such increased population density.

ARTICLE 11 - INSURANCE, LIABILITY AND INDEMNITY

11.1 Tenant's Insurance

The Tenant shall effect and maintain during the Term at its sole cost and expense:

(a) "all risks" insurance upon all property owned by the Tenant or by others and for which property the Tenant is responsible located in the <u>BuildingProject</u> including equipment, furniture, fixtures and Leasehold Improvements in amounts sufficient

- to fully cover, on a replacement cost basis without deduction for depreciation, all such items;
- (b) if applicable, comprehensive form boiler and machinery insurance on a blanket repair and replacement basis with limits for each accident in an amount not less than the full replacement cost of all Leasehold Improvements and all property in the Premises not owned by the Landlord;
- commercial general liability insurance providing coverage no less than that (c) provided under the most recent edition of the IBC 2100 CGL form, including IBC 2313. For the purposes of describing the coverage, and in no way to limit the coverage provided by the noted IBC forms, such coverage shall include coverage on an occurrence basis against claims for bodily injury, personal injury, economic loss, property damage and environmental (with property damage including Building hazards and soil and groundwater contamination being defined in part as physical injury to tangible property and any resultant economic loss) arising from occurrences in or about the BuildingProject or arising from or in any way relating to the Tenant's use or occupancy of the Premises or the Building, Project, and the coverage shall include contractual liability (including coverage of the indemnities provided for in this Lease), and non-owned automobile liability and owner and contractors' protective liability, all in amounts which are from time to time acceptable to a prudent tenant in the community in which the **Building**Project is located (as determined by the Landlord), but not less than \$5,000,000.00 in respect of each occurrence;
- (d) Tenant's legal liability insurance for the full replacement cost of the Premises including loss of the use of the Premises;
- (e) business interruption insurance for a minimum period of 24 months in an amount that will reimburse the Tenant for direct or indirect loss of earnings attributable to all perils insured against in Sections 11.1(a) and 11.1(b) or attributable to prevention of access to the Premises or the BuildingProject as a result of any such perils, including extra expense insurance if applicable;
- (f) pollution legal liability insurance with a limit of \$5,000,000.00 per occurrence in the aggregate; and
- (g) (f) any other form of insurance that the Landlord or any Mortgagee may reasonably require from time to time in form, amounts and for insurance risks acceptable to the Landlord and any Mortgagee.

Should the Tenant fail to maintain any of the insurance required pursuant to this Section 11.1 and should such default continue for two Business Days after Notice to the Tenant, then in addition to any other rights and remedies, the Landlord may, but shall have no obligation to, elect to obtain the required insurance and the Tenant shall upon demand pay to the Landlord, as <u>Additional</u> Rent, the Landlord's cost of obtaining such insurance.

11.2 Form of Tenant Policies

Each policy required pursuant to Section 11.1 shall be in a form and with insurers acceptable to the Landlord, acting reasonably, having reasonable deductibles, and: (a) the insurance described in Sections 11.1(a) and 11.1(b) and any other property damage insurance shall include, as additional insureds (but without liability for premiums) as its interests may appear, the Landlord, any Mortgagee and other Persons with an interest in the BuildingProject from time to time designated in writing by the Landlord provided in each such case there is an insurable interest; (b) the insurance described in Section 11.1(c) shall include as additional insureds (but without liability for premiums) the Landlord, any Mortgagee, any other Persons with an interest in the Building Project from time to time designated in writing by the Landlord and any property manager or facilities manager retained by the Landlord in respect of the BuildingProject; (c) all property damage and liability insurance shall contain provisions for cross-liability and severability of interests among the Landlord, the other insureds and the Tenant; (d) all property damage insurance (including boiler and machinery insurance) shall contain a waiver of any rights of subrogation which the insurer may have against the Landlord and those for whom the Landlord is in law responsible whether the damage is caused by the act, omission or negligence of the Landlord or such other Persons; and (e) shall contain a provision that the Tenant's insurance shall be primary and shall not call into contribution any other insurance available to the Landlord.

11.3 Certified Copies and Notice to Landlord

The Tenant shall provide to the Landlord, prior to the Commencement Date, in digital form satisfactory to the Landlord, certificates of insurance or other evidence satisfactory to the Landlord that the Tenant has obtained all insurance policies required by this Lease and shall provide written evidence of the continuation of such policies not less than ten days prior to their respective expiry dates. Each policy required pursuant to Section 11.1 shall provide that: (a) the insurer must notify the Landlord and any Mortgagee in writing at least 30 days prior to any material change detrimental to the Landlord or any Mortgagee or the cancellation of any such policy; (b) the policy shall not be invalidated in respect of the interests of the Landlord or any Mortgagee or any other additional named insureds by reason of any breach or violation of any warranties, representations, declarations or conditions contained in such policy; and (c) the policy shall be non-contributing with, and shall apply only as primary and not excess to any other insurance available to all and any of the Landlord, any Mortgagee or any other additional named insured referred to above.

11.4 Landlord's Insurance

The Landlord shall effect and maintain during the Term: (a) liability insurance; (b) "all risks" property insurance; (c) boiler and machinery insurance; and (d) such other insurance on the BuildingProject and all property and interest of the Landlord in the BuildingProject as determined by the Landlord, in each case, to the extent, with coverage and in amounts as determined by the Landlord from time to time. The Tenant agrees that notwithstanding the Tenant contributes to the cost of the Landlord's insurance with respect to the BuildingProject, the Tenant shall not have any insurable interest in, or any right to recover any proceeds under any of the Landlord's insurance policies. Without limiting the generality of the foregoing, the Landlord shall be entitled to effect and maintain during the Term, property and business interruption insurance that would provide for, to the extent available on commercially reasonable terms, environmental or other building

accreditation recertification costs, sustainable re-engineering or sustainability design costs incurred after a loss, the incremental costs of debris removal and recycling after a loss, and any additional reconstruction costs associated with reconstruction of the Building to a leading energy conservation and/or sustainability standard such as LEED Existing Buildings: for Building Operations & Maintenance ("EBOMO+M") or BOMA Building Environmental Standards ("BESt") (top quartile or equivalentBEST"), including enhanced external vegetation replacement requirements and the creation or reconstruction of a "green" (vegetative) or "white" (reflective) roof or roofs. In addition, the Landlord may place boiler and machinery breakdown insurance that would permit the replacement of damaged equipment with equipment that increases the building efficiency of the Building or enhances safety, and/or otherwise is consistent with the Environmental Sustainability Management Plan.

11.5 Insurance Risks

The Tenant shall not do, omit to do, or permit to be done or omitted to be done upon the Premises or any other portion of the BuildingProject anything that may contravene or be prohibited by any of the Landlord's insurance policies in force from time to time covering or relevant to any part of the BuildingProject or which would prevent the Landlord from procuring such policies with companies acceptable to the Landlord. If the occupancy of the Premises, the conduct of business in the Premises or any acts or omissions of the Tenant in the Premises or any other portion of the BuildingProject causes or results in any increase in premiums for any of the Landlord's insurance policies, then, without limiting any other rights or remedies of the Landlord, the Tenant shall pay any such increase as Additional Rent forthwith upon receipt of the invoices of the Landlord for such additional premiums. A written report by an Expert concerning the cause of any increase in premiums will be accepted as conclusive evidence of the cause for the purposes of determining the Tenant's liability to pay for increases as Additional Rent.

11.6 Release of Landlord

The Tenant hereby releases the Landlord from any and all claims, actions, causes of action, damages, demands for damages and other liabilities, howsoever arising, that may be made by the Tenant against the Landlord under the provisions of this Lease to the extent of all insurance proceeds paid under the policies of insurance maintained by the Tenant or which would have been paid if the Tenant had maintained the insurance required under this Lease and had diligently processed any claims thereunder. In addition and without limitation, the Tenant agrees that the Landlord, regardless of negligence or alleged negligence on the part of the Landlord or any breach of thethis Lease by the Landlord and, notwithstanding anything else herein contained, shall not be liable for and hereby releases the Landlord from:

- (a) any and all claims, actions, causes of action, damages, demands for damages and other liabilities:
 - (i) for or related to any bodily injury, personal injury, illness or discomfort to or death of the Tenant or any of its agents, officers, contractors, employees, invitees, licensees and any other Person for whom the Tenant is legally responsible in or about the BuildingProject or the Premises including those that may arise due to a Health Emergency or the Landlord's response to it;

- (ii) for or related to any loss or damage to property owned by the Tenant or by others and for which property the Tenant is responsible in or about the BuildingProject or the Premises, and, without limiting the foregoing, the Landlord shall not be liable for any damage caused by steam, water, rain or snow which may leak into, issue or flow from part of the BuildingProject, including the Premises, or from the pipes or plumbing works thereof, or from any other place or for any damage caused by or attributable to the condition or arrangement of any electric or other wiring; and
- (iii) for or related to any closure, quarantine order or decontamination, or environmental investigation and/or remediation (if carried out by the Landlord) whether in contract, tort or on any other basis of liability, statutory or otherwise.
- (b) any loss or damage caused as a result of any damage, destruction, construction, alteration, expansion, expropriation, reduction, repair or reconstruction from time to time of the BuildingProject, any parts or components of the BuildingProject or of improvements on adjoining properties or by anything done or omitted to be done by any other tenant or occupant;
- (c) any act or omission (including theft, malfeasance or negligence) on the part of any agent, contractor or person from time to time employed by Landlord to perform janitorial services, security services, supervision or any other work in or about the Premises or the BuildingProject;
- (d) any loss or damage, however caused, to books of account, records, files, money, securities, negotiable instruments, papers, computer disks, tapes, software, data and other electronic files and their storage media of any kind or to other valuables of the Tenant including art, artworks, statuary, antiques, gems and precious metals of the Tenant and of others;
- (e) any loss or damage arising from obstruction of deliveries to or from the Premises or interruption, cessation, faulty operation, breakdown or failure of any Building Systems, including but not limited to, the supply of any utilities, telecommunication services (whether controlled or owned by the Landlord or not) or other services in, to or serving the <u>BuildingProject</u> or the Premises, whether they are supplied by the Landlord or by others; and
- (f) any indirect or consequential damages including, but not limited to, loss of profit.

11.7 Release of Tenant

The Landlord hereby releases the Tenant, and its agents, officers and employees, and any other Person for whom the Tenant is legally responsible from any liability or claim that may be made by the Landlord against the Tenant under the provisions of this Lease with respect to such loss to the extent of the lesser of: (a) the amount, if any, by which such loss exceeds the amount of insurance the Tenant is required to maintain under the terms of this Lease or actually maintains, whichever is greater; and (b) the proceeds actually paid to the Landlord with respect to such loss under the policies of insurance maintained by the Landlord pursuant to Section 11.4 or which would have been paid if the Landlord had maintained the insurance required under this Lease and had diligently processed any claims thereunder. This release shall be operative only if it is not prohibited by the Landlord's insurance policies and would not place the Landlord in breach of such policies or expose the Landlord to additional costs under or in connection with such policies.

11.8 Indemnity by Tenant

The Tenant shall indemnify and save harmless the Landlord from and against any and all claims, actions, causes of action, damages, demands for damages, losses and other liabilities and expenses (including, without limitation, those in connection with bodily injury (including death), personal injury, illness or discomfort or damage to property including environmental injury, harm and contamination and legal fees on a solicitor and client basis) due to or arising from or out of, subject to Section 11.7, any occurrence in, on or at the Premises or the occupancy or use by the Tenant of the Premises or any other part of the BuildingProject or occasioned wholly or in part by any act or omission of the Tenant, its officers, employees, agents, contractors, invitees, licensees or by any Person (excluding the Landlord and those for whom the Landlord is at law responsible) permitted by the Tenant to be on the Premises or the BuildingProject or due to or arising out of any breach by the Tenant of this Lease.

ARTICLE 12 - ASSIGNMENT, SUBLETTING AND OTHER TRANSFERS

12.1 Transfers

The Tenant shall not enter into, consent to, or permit any Transfer without the prior written consent of the Landlord, which consent shall not be unreasonably withheld but shall be subject to the Landlord's rights under Section 12.2. The Tenant shall pay to the Landlord its then current reasonable charge and all costs incurred (including legal fees and disbursements) in respect of the proposed Transfer. Notwithstanding any statutory provision to the contrary, it shall not be considered unreasonable for the Landlord to withhold its consent if, without limiting any other factors or circumstances which the Landlord may reasonably take into account:

- (a) an Event of Default on the part of the Tenant hereunder has occurred and is continuing;
- (b) the proposed Transfer would be or could result in violation or breach of any covenants or restrictions made or granted by the Landlord to other tenants or occupants, or prospective tenants or occupants, of the BuildingProject;
- (c) in the Landlord's reasonable opinion:
 - (i) either the financial background or the business history and capability of the proposed Transferee is not satisfactory;

- (ii) the nature or character of the proposed business of the proposed Transferee is such that it might harm the Landlord's business or reputation or reflect unfavourably on the BuildingProject, or the image of any of them, or is unethical, immoral or illegal;
- (iii) the use of the Premises by the proposed Transferee could be incompatible with the other businesses or activities being carried on in the BuildingProject or could result in excessive demands being placed on the Building Systems or other Common Areas and Facilities;
- (iv) if the Transfer affects less than all of the Premises, the portion affected or the portion remaining are not acceptable in respect of size, access or configuration; or
- (v) the use of the Premises by the proposed Transferee may imperil any existing or intended certification or accreditation of the BuildingProject (or any part thereof) or performance targets for the BuildingProject (or any part thereof) as are set out in Schedule "E", of this Lease (including Sections 1.2 and 1.3 herein.thereof);
- (d) the proposed Transferee or any principal of the proposed Transferee or any principal shareholder of the proposed Transferee has a history of defaults under other commercial leases or does not have a satisfactory history of compliance with laws;
- (e) the Landlord at the time has, or will have in the next ensuing three month period, other premises in the <u>BuildingProject</u> suitable for leasing to the proposed Transferee;
- (f) the Basic Rent and Additional Rent payable by the Transferee is less than the Basic Rent and Additional Rent payable by the Tenant hereunder as at the effective date of the Transfer except in the case where the Landlord determines, in its sole discretion, that payment of lesser rent by the Transferee will not detrimentally affect the leasing program for the BuildingProject; or
- (g) the proposed Transfer is to: (i) an existing tenant or occupant of the BuildingProject or of any other building owned or managed by the Landlord or any of its affiliates within the same market area as determined by the Landlord; or (ii) a consulate, embassy, trade commission or other representative of a foreign government; or (iii) a government, quasi-government or public agency, service or office.

Any consent by the Landlord to a Transfer shall not constitute a waiver of the necessity for such consent to any subsequent Transfer.

12.2 Tenant's Notice, Landlord's Right to Terminate

If the Tenant intends to effect a Transfer, the Tenant shall give prior Notice to the Landlord of such intent specifying the identity of the Transferee, the type of Transfer contemplated, the part of the Premises affected and the financial and other terms of the Transfer, and shall provide such financial, business or other information relating to the proposed Transferee and its principals as the Landlord or any Mortgagee reasonably requires, together with copies of all documents which record the particulars of the proposed Transfer. The Landlord shall, within 15 days after having received such Notice and all requested information, notify the Tenant either that:

- (a) it consents or does not consent to the Transfer in accordance with the provisions of this Lease; or
- (b) it elects to terminate this Lease as to the part of the Premises affected by the proposed Transfer, or as to the whole Lease and Premises if the proposed Transfer affects all of the Premises.

If the Landlord elects to terminate this Lease it shall stipulate in its Notice the termination date of this Lease, which date shall be the date of possession contemplated under the proposed Transfer (provided that if such date is less than 30 days following the giving of Notice of such election, the Landlord may elect to have the termination date 30 days following the giving of Notice). If the Landlord elects to terminate this Lease, the Tenant may notify the Landlord within ten days following receipt of such Notice of the Tenant's intention to refrain from such Transfer and, if the Tenant provides such Notice within such time period, then the Landlord's election to terminate this Lease shall become void. If the Tenant fails to deliver such Notice within such time period, then this Lease shall, as to the whole or affected part of the Premises, as the case may be, be terminated on the date of termination stipulated by the Landlord in its Notice of election to terminate. If the Tenant is required to deliver possession of a part only of the Premises, the Tenant shall pay all costs incurred in connection with rendering that part functionally separate and suitable for separate use and occupancy, including partitioning and providing entrances and services.

12.3 Conditions of Transfer

The following terms and conditions apply in respect of a Transfer:

- (a) the Tenant and the Transferee shall execute, prior to the Transfer being made, an agreement with the Landlord in the Landlord's form including the Transferee's covenant to be bound by all of the terms of this Lease;
- (b) notwithstanding any Transfer, the Tenant shall remain liable under this Lease and shall not be released from performing any of the terms of this Lease. The Tenant's liability shall continue notwithstanding any amendment of this Lease throughout the Term and any exercise of any renewal or extension of the Term provided for herein, regardless of whether or when an amendment of this Lease is made (however the original Tenant's liability will not be increased by any amendment that it is not a party to) and notwithstanding that the Landlord may collect Rent from the Transferee:

- (c) if the Basic Rent and Additional Rent (net of reasonable out of pocket costs for commissions, for cash allowances and for Alterations required by and made for the Transferee by the Tenant, amortized on a straight line basis over the term of the Transfer) to be paid by the Transferee under such Transfer exceeds the Basic Rent and Additional Rent payable by the Tenant hereunder, the amount of such excess shall be paid by the Tenant to the Landlord. If the Tenant receives from any Transferee, either directly or indirectly, any consideration other than Basic Rent or Additional Rent for such Transfer, either in the form of cash, goods or services, the Tenant shall immediately pay to the Landlord an amount equivalent to such consideration;
- (d) if the Transfer is a sublease, the Transferee will agree to waive any statutory or other right to apply to a court or to otherwise elect to: (i) retain the unexpired term of thethis Lease or the unexpired term of the sublease; (ii) obtain any right to enter into any lease or other agreement directly with the Landlord; or (iii) otherwise remain in possession of any portion of the Premises, in any case where thethis Lease is terminated, surrendered or otherwise cancelled, including, without limitation, any disclaimer, repudiation, surrender or other termination (each of these transactions being referred to as an "Early Termination") by any trustee in bankruptcy of the Tenant or a Transferee, by any court appointed officer, or by the Tenant or a Transferee in connection with any insolvency proceedings;
- (e) if there is an Early Termination, the Tenant and any Transferee (except the bankrupt or insolvent Tenant or Transferee) to whom the Landlord gives Notice within 60 days after the Early Termination, shall be considered to have entered into a lease with the Landlord on the same terms and conditions as are contained in this Lease except that the term of the lease shall commence on the date of the Early Termination and shall expire on the date this Lease would have expired but for the Early Termination; and
- (f) notwithstanding the effective date of any permitted Transfer as between the Tenant and the Transferee, all Rent for the month in which such effective date occurs shall be paid in advance by the Tenant so that the Landlord will not be required to accept partial payments of Rent for such month from either the Tenant or the Transferee.

12.4 Corporate Records

Upon the Landlord's request, the Tenant shall: (a) deliver a statutory declaration by one of its senior officers setting forth the details of its corporate and capital structure; and (b) make available to the Landlord or its representatives all of its corporate or partnership records, as the case may be, for inspection at all reasonable times, in order to ascertain whether any Change of Control has occurred.

12.5 Permitted Transfers

Notwithstanding Section 12.1 and provided that the Required Conditions are <u>at all times</u> satisfied and there is not then an Event of Default, the Tenant shall have the right on prior Notice to the

Landlord, but without being required to obtain the Landlord's consent, to effect a Transfer in compliance with Section 12.3 in favour of a Permitted Transferee and the Landlord's right to terminate this Lease pursuant to Section 12.2 shall not apply to such a Transfer.

12.6 No Advertising

The Tenant shall not advertise <u>nor otherwise disclose through any medium</u> that the whole or any part of the Premises are available for a Transfer and shall not permit any broker or other Person to do so, <u>and shall not permit any marketing</u>, <u>advertisement or other disclosure in any media format</u>, unless the text and format of such advertisement is approved in writing by the Landlord. No such <u>marketing</u>, advertisement <u>or other disclosure</u> shall contain any reference to the rental rate of the Premises <u>or any other financial terms pertaining to the Premises</u>, this <u>Lease or the proposed</u> Transfer.

12.7 Sales or Dispositions by Landlord

The Landlord shall have the unrestricted right to sell, transfer, lease, license, charge or otherwise dispose of all or any part of its interest in the BuildingProject or any interest of the Landlord in this Lease. In the event of any sale, transfer, lease or other disposition the Landlord shall thereupon, and without further agreement, be released of all liability under this Lease arising from and after such disposition. If required by the Landlord in connection with any sale, transfer, charge or other disposition the Tenant shall, within five Business Days of request, provide to the Landlord, prospective purchasers and Mortgagees and their respective agents and consultants, access to the current financial statements of the Tenant and any Indemnifier. If the Tenant is listed on a recognized stock exchange in Canada or the United States, the Tenant agrees to provide instead copies of the Tenant's annual reports, quarterly reports and all other publicly distributed reporting materials.

ARTICLE 13 - LANDLORD FINANCING AND STATUS CERTIFICATES

13.1 Subordination and Postponement

- (a) This Lease and the rights of the Tenant in this Lease shall be subject and subordinate to any and all Mortgages and the Tenant, on request by and without cost to the Landlord, shall, within five Business Days after such request, execute and deliver any and all instruments required by the Landlord to evidence such subordination. Upon request by the Tenant at the time of any request for confirmation of subordination, the Landlord shall make reasonable commercial efforts to obtain from any Mortgagee an acknowledgement and assurance in writing addressed to the Tenant, whereby such Mortgagee acknowledges that, in the event of any such Mortgagee realizing upon the security, it will not disturb the Tenant and will permit the Tenant to remain in possession under this Lease in accordance with its terms, so long as the Tenant is not in default.
- (b) The Landlord, as to any Mortgage, and a Mortgagee, as to any Mortgage held by it, may, by Notice to the Tenant, elect that this Lease and the rights of the Tenant hereunder shall be prior to such Mortgage(s) and the Tenant, on request by and

without cost to the Landlord, shall, within five Business Days after such request, execute and deliver any and all instruments required by the Landlord or the Mortgagee, as the case may be, to confirm priority to this Lease over the Mortgage(s).

13.2 Attornment

At any time after any of the following has occurred:

- (a) if a Mortgagee delivers a Notice of attornment;
- (b) if a Mortgagee shall take possession of the **Building**Project or the Premises; or
- (c) if the interest of the Landlord is transferred to any Person (in this Article referred to as a "Purchaser") by reason of foreclosure or other proceedings for enforcement of any Mortgage, or by delivery of a conveyance,

the Tenant shall, at the option of the Mortgagee or the Purchaser, as the case may be, exercisable by Notice in writing to the Tenant, be deemed to have attorned to the Mortgagee or the Purchaser, as the case may be, upon receipt of such Notice. The Landlord, the Mortgagee or the Purchaser, as the case may be, may require the Tenant to enter into all instruments required by the Landlord, the Mortgagee or the Purchaser, as the case may be, to confirm such attornment. Upon such attornment the obligations of the Tenant under this Lease shall continue in full force and effect upon all the same terms, conditions and covenants in this Lease.

13.3 Status Certificates

The Tenant shall at any time and from time to time execute and deliver to the Landlord, or as the Landlord, a Mortgagee or a Purchaser may direct, within five Business Days after it is requested, a certificate of the Tenant, in the form supplied, addressed to the Landlord, the Mortgagee or the Purchaser, as the case may be, and/or any prospective purchaser, lessor or Mortgagee, certifying such particulars, information and other matters in respect of the Tenant, the Premises and this Lease that the Landlord, the Mortgagee or the Purchaser, as the case may be, may request.

13.4 Reliance

Notwithstanding that a Mortgagee or a Purchaser is not a party to this Lease, it shall be entitled to rely upon and enforce the provisions of this Lease which are stated to be for its benefit and, without limitation, the Mortgagee shall be entitled to act as agent for the Landlord to the extent necessary to enforce any such provisions.

ARTICLE 14 - DAMAGE, DESTRUCTION

14.1 Damage to Premises

If there is damage, destruction or contamination by the presence of a Hazardous Substance or otherwise (collectively "Damage") to all or any material part of the Premises such that the Premises are rendered untenantable or inaccessible, then:

- (a) if in the reasonable opinion of the Expert, the Damage can be substantially repaired under Applicable Laws within 180 days from the date of such Damage (employing normal construction methods without overtime or other premium), the Landlord shall forthwith repair such Damage other than Damage to Leasehold Improvements and any other property that is not the responsibility of or is not owned by Landlord; and
- (b) if in the reasonable opinion of the Expert, the Damage cannot be substantially repaired under Applicable Laws within 180 days from the date of such Damage (employing normal construction methods without overtime or other premium), then:
 - (i) the Landlord may elect to terminate this Lease as of the date of such casualty by Notice delivered to the Tenant not more than 20 days after receipt of the Expert's opinion; and
 - (ii) if such Damage occurs during the last two Lease Years, either party may elect to terminate this Lease as of the date of such Damage by Notice delivered to the other not more than 20 days after receipt of the Expert's opinion,

failing which the Landlord shall forthwith repair such Damage other than Damage to Leasehold Improvements or property that is not the responsibility of or is not owned by Landlord.

14.2 Abatement

If the Landlord is required to repair Damage to the Premises under Section 14.1 the Basic Rent payable by the Tenant shall be proportionately reduced to the extent that the Premises are rendered untenantable or inaccessible, from the date of the Damage until 30 days after completion by the Landlord of the repairs to the Premises or until the Tenant again uses the Premises (or the part thereof rendered untenantable), whichever first occurs. The Tenant shall effect its own repairs as soon as possible after completion of the Landlord's repairs in accordance with its obligations under this Lease. Notwithstanding the foregoing, there shall be no abatement or reduction of Basic Rent where the Landlord's repairs to the Premises take less than ten days to complete after the damage occurs.

14.3 Termination Rights

Notwithstanding anything else contained in this Lease, if: (a) there is Damage within or to the Building so as to affect 25% or more of the Rentable Area of the Building; or (b) in the reasonable opinion of the Expert the Building is unsafe or access or services are affected and, in either case, cannot be substantially repaired under Applicable Laws within 180 days from the date of such Damage (employing normal construction methods without overtime or other premium); or (c) the proceeds of insurance are substantially insufficient to pay for the costs of repair or rebuilding or are not payable to or received by the Landlord; or (d) Damage is caused by an occurrence against which the Landlord is not insured or beyond the extent to which the Landlord is required to insure under this Lease; or (e) any Mortgagee(s) or other Person entitled to the insurance proceeds shall not consent to the repair and rebuilding, then the Landlord may terminate this Lease by giving to the Tenant Notice of such termination within 60 days of the Damage, in which event the Term shall cease and be at an end as of the date of such Damage and the Rent and all other payments for which the Tenant is liable under the terms of this Lease shall be apportioned and paid in full to the date of termination (subject to any abatement under Section 14.2).

14.4 Landlord's Rights on Rebuilding

In the event of Damage to the Building and if this Lease is not terminated in accordance with Sections 14.1 or 14.3, the Landlord shall forthwith repair such Damage, but only to the extent of the Landlord's obligations under the terms of the various leases for premises in the Building (including this Lease) and exclusive of any tenant's responsibilities with respect to such repair. In repairing or rebuilding the Building or the Premises the Landlord may use drawings, designs, plans and specifications other than those used in the original construction and may alter or relocate the Building on the Lands, the Common Areas and Facilities or any part thereof, and may alter or relocate the Premises, provided that the Building as repaired or rebuilt is in compliance with all Applicable Laws and is of at least a similar standard, and the Premises as altered or relocated shall be of approximately the same size as the original Premises.

ARTICLE 15 - DEFAULT AND REMEDIES

15.1 Events of Default

Any of the following constitutes an "Event of Default" under this Lease:

- (a) any Rent is in arrears and is not paid within five days after Notice from the Landlord;
- (b) the Tenant has breached any of its obligations in this Lease and, if such breach is capable of being remedied and is not otherwise listed in this Section 15.1, after Notice from the Landlord:
 - (i) the Tenant fails to remedy such breach within ten days (or such shorter period as may be provided in this Lease); or

- (ii) if such breach cannot reasonably be remedied within ten days or such shorter period, the Tenant fails to commence to remedy such breach within such ten days or shorter period or thereafter fails to proceed diligently to remedy such breach;
- (c) thethis Lease or any goods, chattels or equipment of the Tenant is seized, taken or exigible in execution or in attachment or if a writ of execution or enforcement is issued against the Tenant and such writ is not stayed or vacated within ten days after the date of such issue:
- (d) the Tenant or any Indemnifier becomes insolvent or commits an act of bankruptcy or takes the benefit of any statute for bankrupt or insolvent debtors or makes any proposal, assignment, compromise or arrangement with its creditors, or if a receiver is appointed for all or part of the business, property, affairs or revenues of the Tenant:
- (e) the Tenant makes a bulk sale of its goods (other than in conjunction with a Transfer approved by the Landlord) or moves or commences, attempts or threatens to move its goods, chattels and equipment out of the Premises (other than in the normal course of its business);
- (f) the Tenant abandons or attempts to abandon the Premises or ceases to conduct business from the Premises, or the Premises become vacant or substantially unoccupied for a period of ten consecutive days; or
- (g) the Tenant purports to effect a Transfer other than in compliance with the provisions of this Lease; or
- (h) three or more Events of Default under this Lease have occurred within any two year period.

15.2 Remedies

If and whenever an Event of Default occurs, the Landlord shall have the following rights and remedies, exercisable immediately and without further Notice and at any time while the Event of Default continues:

(a) to terminate this Lease and re-enter the Premises. The Landlord may remove all Persons and property from the Premises and store such property at the expense and risk of the Tenant or sell or dispose of such property in such manner as the Landlord sees fit without Notice or liability to the Tenant. Notwithstanding any termination of this Lease, the Landlord shall be entitled to receive Rent and all Rental Taxes up to the time of termination plus accelerated Rent as provided in this Lease and damages including, without limitation: (i) damages for the loss of Rent suffered by reason of this Lease having been prematurely terminated; (ii) costs of reclaiming, repairing and re-leasing the Premises; and (iii) legal fees and disbursements on a solicitor and client basis;

- to enter the Premises as agent of the Tenant and to relet the Premises for whatever (b) length of time and on such terms as the Landlord in its discretion may determine including, without limitation the right to: (i) take possession of any property of the Tenant on the Premises; (ii) store such property at the expense and risk of the Tenant; (iii) sell or otherwise dispose of such property in such manner as the Landlord sees fit and the Tenant shall indemnify the Landlord against any claims from any third parties including the Tenant's lenders; and (iv) make alterations to the Premises to facilitate the reletting. The Landlord shall receive the rent and proceeds of sale as agent of the Tenant and shall apply the proceeds of any such sale or reletting first, to the payment of any expenses incurred by the Landlord with respect to any such reletting or sale, second, to the payment of any indebtedness of the Tenant to the Landlord other than Rent and third, to the payment of Rent in arrears, with the residue to be held by the Landlord and applied to payment of future Rent as it becomes due and payable. The Tenant shall remain liable for any deficiency to the Landlord;
- (c) to remedy or attempt to remedy the Event of Default for the account of the Tenant and to enter upon the Premises for such purposes. The Landlord shall not be liable to the Tenant for any loss, injury or damages caused by acts of the Landlord in remedying or attempting to remedy the Event of Default. The Tenant shall pay to the Landlord, on demand, all expenses incurred by the Landlord in remedying the Event of Default, together with an administration fee of 15% and interest at the Default Rate from the date such expense was incurred by Landlord;
- (d) to recover from the Tenant all damages, costs and expenses incurred by the Landlord as a result of the Event of Default including any deficiency between those amounts which would have been payable by the Tenant for the portion of the Term following such termination and the net amounts actually received by the Landlord during such period of time with respect to the Premises; and
- (e) to recover from the Tenant the full amount of the current month's Rent together with the next three months' instalments of Rent, which shall immediately become due and payable as accelerated rent.

15.3 Distress

Notwithstanding any provision of this Lease or any provision of any present or future Applicable Laws, none of the goods, chattels or trade fixtures on the Premises at any time during the Term shall be exempt from levy by distress for Rent in arrears, and the Tenant waives any such exemption. If the Landlord makes any claim against the goods and chattels of the Tenant by way of distress this provision may be pleaded as an estoppel against the Tenant in any action brought to test the right of the Landlord to levy such distress.

15.4 Interest and Costs

The Tenant shall pay to the Landlord upon demand: (a) interest at the Default Rate on all Rent required to be paid hereunder from the due date for payment until fully paid and satisfied; and (b)

the Landlord's then current reasonable administration charge for each Notice of default given by the Landlord to the Tenant under this Lease. The Tenant shall pay and indemnify the Landlord against damages, costs and expenses (including, without limitation, all legal fees on a solicitor and client basis) incurred in enforcing the terms of this Lease, or with respect to any matter or thing which is the obligation of the Tenant under this Lease, or in respect of which the Tenant has agreed to insure or to indemnify the Landlord.

15.5 Remedies Cumulative

No reference to or exercise of any specific right or remedy by the Landlord shall prejudice or preclude the Landlord from exercising or invoking any other remedy, whether allowed under this Lease or generally at law or in equity, and the express provisions of this Lease as to certain rights and remedies are not to be interpreted as excluding any other or additional rights and remedies available to the Landlord generally at law or in equity.

ARTICLE 16 - MISCELLANEOUS

16.1 Relationship of Parties

Nothing contained in this Lease shall create any relationship between the parties other than that of landlord and tenant, and, without limitation, nothing in this Lease shall be construed to constitute the Landlord and the Tenant as partners, joint venturers or members of a joint or common enterprise.

16.2 Consent Not to be Unreasonably Withheld

Except as otherwise specifically provided in this Lease, the Landlord and the Tenant, and each Person acting for them, in granting a consent or approval or making a determination, designation, calculation, estimate, conversion or allocation under this Lease, will act reasonably and in good faith and each Expert or other professional Person employed or retained by the Landlord will act in accordance with the applicable principles and standards of such Person's profession. The Tenant's sole remedy against the Landlord in respect of any breach or alleged breach of this Section shall be an action for specific performance and, without limitation, the Landlord shall not be liable for damages and the Tenant shall not be entitled to any other rights or remedies. If either party withholds any consent or approval where it is required to act reasonably, such party shall, on written request, deliver to the other party a written statement giving the reasons for withholding the consent or approval.

16.3 Overholding

The Tenant has no right to remain in possession of the Premises after the end of the Term. If the Tenant remains in possession of the Premises after the end of the Term or any extension term(s) with the consent of the Landlord but without entering into a new lease or other agreement then, notwithstanding any statutory provisions or legal presumption to the contrary, there shall be no tacit renewal of this Lease or the Term or any extension term(s) and the Tenant shall be deemed to be occupying the Premises as a tenant from month to month (with either party having the right to terminate such month to month tenancy at any time on 30 days' Notice, whether or not the date

of termination is at the end of a rental period) at a monthly Basic Rent payable in advance on the first day of each month equal to 150% of the monthly amount of Basic Rent payable during the last month of the Term and otherwise upon the same terms, covenants and conditions as in this Lease insofar as these are applicable to a monthly tenancy and, for greater certainty, including liability for all Additional Rent.

57

16.4 Registration

Neither the Tenant nor anyone on the Tenant's behalf or claiming under the Tenant (including any Transferee) shall register this Lease or any Transfer against the Lands. The Tenant may register a Notice on title to the Landlord a short form, notice or caveat of this Lease provided that: (a) a copy of thethis Lease is not attached; (b) no financial terms are disclosed; (c) the Landlord gives its prior written approval to the short form, notice or caveat; and (d) the Tenant pays the Landlord's reasonable costs on account of the matter. The Landlord may limit such registration to one or more parts of the Lands. Upon the expiration or other termination of the Term the Tenant shall immediately discharge or otherwise vacate any such short form, Notice or caveat. If any part of the Lands which in the opinion of the Landlord are surplus is transferred, the Tenant shall forthwith at the request of the Landlord discharge or otherwise vacate any such short form, Notice or caveat as it relates to such part. If any part of the Lands are made subject to any easement, right-of-way or similar right, the Tenant shall immediately at the request of the Landlord postpone its registered interest to such easement, right-of-way or similar right.

16.5 Unavoidable Delay

If any party is bona fide delayed, or hindered in or prevented from the performance of any term, covenant or act required by this Lease by reason of any cause beyond the control of the party affected including, without limitation, strikes, lockouts or other labour disputes, the enactment, amendment or repeal of any Applicable Laws, the failure of any existing tenant or occupant to vacate the Premises, shortages or unavailability of labour or materials, riots, insurrection, sabotage, rebellion, war, acts of terrorism, act of God, Health Emergency or any other similar reason ("Unavoidable Delay"), then performance of such term, covenant or act is excused for the period of the delay and the party so delayed, hindered or prevented shall be entitled to perform such term, covenant or act within the appropriate time period after the expiration of the period of such delay. However, the provisions of this Section do not operate to excuse the Tenant from the prompt payment of Rent and any other payments required by this Lease and Unavoidable Delay shall not include any delay caused by the parties' default or act or omission, any delay avoidable by the exercise of reasonable care by such party or any delay caused by lack of funds of such party. The Landlord shall also be excused from the performance of any term, covenant or act required hereunder if the performance of such item would be in conflict with any directive or policy of any governmental or quasi-governmental authority having jurisdiction over the **Building**Project in respect of any energy, conservation, health, safety or security matter.

16.6 Decisions of Experts

The decision of any Expert whenever provided for under this Lease and any certificate of an Expert in each case addressed to both parties shall be final and binding on the parties and there shall be no further right of dispute or appeal.

16.7 Notices

Any notice, demand, statement or request ("Notice") required or permitted to be given under this Lease shall be in writing and shall be deemed to have been duly given if personally delivered, delivered by courier or mailed by registered prepaid post, in the case of Notice to the Landlord, to it at the address set out in Section 1.1(a)(ii) and in the case of Notice to the Tenant, to it at the Premises. Notice may not be given by facsimile transmission, Notwithstanding anything to the contrary, no electronic communications of Notices (other than routine operational notices, such as requests for after-hours services (such as, by way of example, HVAC services or entry notices) will be considered as legally effective, it being understood that Notices issued pursuant to Article 15 of this Lease, termination or extension (or renewal) notices, or any other notices of a material nature will not be considered legally effective if delivered electronically. Further, notwithstanding the foregoing, during a Health Emergency, the parties may deliver any and all Notices by electronic mail or any other electronic communication.

Any such Notice given in accordance with the above requirements shall be deemed to have been given, if mailed, on the fifth day following the date of such mailing or, if delivered, on the day on which it was delivered so long as such delivery was prior to 5:00 p.m. on a Business Day (and, if after 5:00 p.m. or if any such day is not a Business Day, then it shall be deemed to have been delivered on the next Business Day). Either party may from time to time by Notice change the address to which Notices to it are to be given. Notwithstanding the foregoing, during any interruption or threatened interruption in postal services, any Notice shall be personally delivered or delivered by courier. If a copy of any Notice to the Tenant is to be sent to a second address or to another Person other than the Tenant, the failure to give any such copy shall not vitiate the delivery of the Notice to the Tenant.

16.8 Confidentiality, Personal Information

The Tenant shall keep confidential all financial information in respect of this Lease, provided that it may disclose such information to its auditors, consultants and professional advisors so long as they have first agreed to respect such confidentiality. Any Tenant or Indemnifier that is an individual person consents to the collection and use of their personal information, as provided directly or collected from third parties, for the purposes of the Landlord considering the Tenant's offer to lease (preceding this lease), any right under this Lease, any renewal, extension or early surrender of this Lease, and determining the suitability of the Tenant or Indemnifier, as applicable, (both initially and on an on-going basis), including the disclosure of such information to existing and potential lenders, investors and purchasers. The Landlord may from time to time designate certain information in respect of the EnvironmentalSustainability Management Plan and the compliance of the BuildingProject or the Premises with the EnvironmentalSustainability Management Plan, as confidential, provided that the Tenant may, on prior written notice to the Landlord, disclose such environmental information as necessary to its auditors, consultants and professional advisors so long as they have first agreed to respect such confidentiality, or unless otherwise required pursuant to Applicable LawLaws.

16.9 Power, Capacity and Authority

The Landlord and the Tenant covenant, represent and warrant to each other that they have the power, capacity and authority to enter into this Lease and to perform its obligations hereunder and that there are no covenants, restrictions or commitments given by it which would prevent or inhibit it from entering into this Lease.

16.10 Liability of Landlord

Any liability of the Landlord under this Lease shall be limited to its interest in the <u>BuildingProject</u> from time to time. If the Landlord consists of more than one Person, the liability of each such Person shall be several and be limited to its percentage interest in the <u>BuildingProject</u>.

SCHEDULES

- "A" Building Specific Information
- "B" Sketch Showing Premises
- "C" Rules and Regulations
- "D" Indemnity Agreement
- "E" Environmental Sustainability Management Plan

IN WITNESS WHEREOF the parties hereto have executed this Lease.

| (Landlord) |
|--|
| Per: Name: Title: c/s |
| Per: Name: Title: I/We have authority to bind the Corporation. |
| <*> (Tenant) |
| Per: Name: Title: |
| Per: Name: Title: I/We have authority to bind the Corporation. |
| <*> (Indemnifier) |
| Per: Name: Title: c/s |
| Per: Name: Title: I/We have authority to bind the Corporation. |

SCHEDULE A BUILDING- SPECIFIC INFORMATION

Part 1: Legal Description:

Part 2: Business Hours:

Part 3: Measurement Standard: <u>ANSI/BOMA Z65.1 - 2017</u>

SCHEDULE B SKETCH SHOWING PREMISES

SCHEDULE C RULES AND REGULATIONS

1. Security and Safety

- (a) The Landlord may from time to time adopt appropriate systems and procedures for the security and safety of the BuildingProject and the tenants and occupants and contents thereof, and the Tenant shall comply with the Landlord's reasonable requirements in respect of such systems and procedures.
- (b) The Tenant shall participate in fire drills and evacuations of the Building as directed by the Landlord. In the event of an emergency, the Tenant shall vacate the Building if the Landlord or any public authority so directs in the manner prescribed by the Landlord or such public authority.
- (c) The Tenant shall not keep any inflammable oils or other inflammable, dangerous, corrosive or explosive materials in the Premises or the BuildingProject, save and except for incidental amounts used in the Tenant's business operations and kept and used in accordance with all Applicable Laws.
- (d) The Tenant shall not bring any weapons, including firearms, knives (except normal kitchen utensils and office equipment such as scissors), fireworks or other similar implements, into the Premises or the BuildingProject at any time.
- (e) The Tenant shall use and encourage the use of <u>proper hand washing and</u> alcohol based waterless hand sanitizers in all washroom and food preparation facilities within the Premises.

2. Use of Premises

- (a) The Tenant shall not use or permit the Premises to be used for residential, lodging or sleeping purposes, or for the storage of personal effects or articles not required for business purposes.
- (b) The Tenant shall not cook or heat any foods or liquids (other than the heating of food in microwave ovens or the heating of water or coffee in coffee makers or kettles) in the Premises without the written consent of the Landlord, and shall not permit on the Premises the use of equipment for dispensing food or beverages or for the preparation, solicitation of orders for, sale, serving or distribution of food or beverages.
- (c) Only Persons approved from time to time by the Landlord may solicit orders for, sell, serve or distribute foods or beverages in the <u>BuildingProject</u> or use the entrances, elevators or corridors for any such purpose.

3. **Operation of Premises**

- (a) The Tenant shall place all refuse and recyclables in the receptacles provided by the Tenant in the Premises or in the receptacles (if any) provided by the Landlord for the BuildingProject, and shall otherwise keep the Lands and the Building and Project (including) the sidewalks and driveways outside the Building) free of all refuse.
- (b) The Tenant shall participate in all <u>Building</u> recycling, energy reduction and water conservation programs <u>for the Project</u> as may be determined by the Landlord from time to time.
- (c) The Tenant shall neither obstruct nor use the entrances, passages, escalators, elevators and staircases of the Building or the sidewalks and driveways outside the Building for any purpose other than ingress to and egress from the Premises and the Building.
- (d) Where possible, the Tenant shall use compact florescent light bulbs in portable indirect lighting, provided such compact florescent bulbs are of low mercury content and suitable for landfillLED lights.
- (e) The Landlord shall be entitled to refuse to collect refuse and recyclables if not properly sorted into the appropriate recyclable container, and the Landlord shall be entitled to charge the Tenant for any costs it incurs as a result of the Tenant's failure to comply with the Building recycling program for the Project.
- (f) The Tenant shall properly separate, store, and dispose of hazardous waste they produce (such as, by way of example, electronic waste) as relevant, from other refuse and recyclables, in accordance with the Landlord's directives and all provincial or local requirements.

4. Repair, Maintenance, Alterations and Improvements

The Tenant shall carry out the Tenant's repairs, maintenance, Alterations and improvements in the Premises consistent with <u>thethis</u> Lease, the <u>EnvironmentalSustainability</u> Management Plan and the Tenant Construction Manual, only during times agreed to in advance by the Landlord and in a manner which will not interfere with the rights of other tenants <u>inof</u> the <u>Building</u>Project.

5. **Deliveries**

The Tenant shall not make or receive any deliveries from or to the Premises except through the entrances, elevators and corridors and at the times designated by the Landlord.

6. Articles

(a) Any furniture or equipment being moved in or out of the Premises by the Tenant shall be moved through the entrances, elevators and corridors and at the times

designated by the Landlord. All appliances used to move articles in or out of the Premises shall be equipped with rubber tires, slide guards and any other safeguards required by the Landlord.

- (b) The Tenant shall not place in or move about the Premises any heavy machinery or equipment or anything liable to injure or destroy any part of the Premises or the BuildingProject without the prior written consent of the Landlord.
- (c) The Tenant may be required to report to the <u>Buildingproperty</u> manager <u>for the Project</u> as to whether items of equipment purchased by the Tenant for use within the Premises are <u>Energy StarENERGY STAR</u>, EcoLogo^M, Green SealTM or otherwise approved by a credible authority (as determined by the Landlord acting reasonably) as environmentally friendly.
- (d) [OPTIONAL: The Tenant shall not permit the use of, within its the Premises, any stand _alone energy intensive equipment designed to modify indoor air temperature or humidity, such as portable air conditioners, space heaters, humidifiers or dehumidifiers.]

7. Windows

The Tenant shall not install curtains, blinds or other window coverings without the prior written consent of the Landlord. Window coverings that are installed by the Tenant shall comply with any uniform scheme of the Building. If the Building has operable windows, the Tenant will keep such windows closed during any smog alert days.

8. Washrooms and Water Fixtures

The Tenant shall be permitted to use those washrooms on the floor of the Building on which the Premises are situated or, in lieu thereof, those washrooms designated by the Landlord from time to time. The Tenant shall not use the washrooms or other water fixtures for any purposes other than those for which they were intended, and no sweepings, rubbish, rags, ashes or other substances shall be thrown into them.

9. Locks and Security Systems

The Landlord may from time to time install and change locks and/or security systems on entrances to the Premises and the Building. The Tenant shall be supplied with a reasonable number of keys or other entry devices for each installation. Any additional keys or entry devices required by the Tenant must be obtained from the Landlord at the Tenant's expense. The Tenant shall not place or cause to be placed any additional locks or security systems on entrances to the Premises without the prior written consent of the Landlord. At the end of the Term, the Tenant shall return to the Landlord all keys and other entry devices for the Premises and the Building which are in the possession of the Tenant. If without the Landlord's consent, the Tenant installs lock(s) incompatible with the Building master locking system:

- (a) <u>the Landlord</u>, without abatement of Rent, shall be relieved of any obligation under <u>thethis</u> Lease to provide any service to the affected areas which require access thereto;
- (b) <u>the Tenant shall indemnify the Landlord against any expense as a result of forced entry thereto which may be required in an emergency; and</u>
- (c) <u>the Landlord's may at any time remove such lock(s) at the Tenant's expense.</u>

10. Bicycles and Vehicles

The Tenant shall not bring any bicycles or other vehicles within any part of the Lands or BuildingProject except in such area or areas designated by the Landlord from time to time.

11. Living Creatures

The Tenant shall not bring any living creatures, including animals (except <u>service</u> dogs <u>assisting</u> the <u>disabled</u>), reptiles, insects, birds or fish within any part of the <u>Lands or Building Project</u> without the consent of the Landlord.

12. Indoor Plants and Vegetation, and Pest Control

The Tenant shall ensure that no pesticides or herbicides are used within the Premises. The Tenant shall maintain any indoor plants and vegetation within the Premises in a healthy state, provided that any fertilizers used shall meet EcoLogo^M, Green SealTM or equivalent standards.

13. Antennae, Satellite Dish

The Tenant shall not install any radio or television antenna or satellite dish on any part of the Lands or Building Project without the prior written consent of the Landlord, which may include execution of the Landlord's standard form of license agreement.

14. **Smoking**

The Tenant shall not permit smoking in <u>or on</u> any part of the <u>BuildingProject</u>, including the Premises.

15. Canvassing, Soliciting and Peddling

Canvassing, soliciting and peddling in or about the Lands and Building are Project is prohibited.

16. Employees, Agents and Invitees

In these Rules and Regulations, "Tenant" includes the employees, agents, invitees and licensees of the Tenant and others permitted by the Tenant to use or occupy the Premises.

17. **Health Screening Amenity Facility**

The Landlord shall be entitled, during such time as there is a Health Emergency or a Health Emergency Plan is in effect, to require all occupants to comply with reasonable measures imposed in respect thereof by the Landlord, including health screening, the use of hand washing and other sanitation products directly related to the management of the health threat, attendance at mandatory training sessions, and the use of additional protective clothing by all occupants, invitees and tenants such as protective barriers, gloves and masks to establish Rules and Regulations in regards to any Amenity Facility for the Project with respect to access and its proper use and operation of the Amenity Facility and the Tenant, and those for whom the Tenant is at law responsible (including its employees), shall at all times comply with such Rules and Regulations. Such Rules and Regulations shall be read as forming part of this Lease and the Tenant, and those for whom it is in law responsible (including its employees), who wish to utilize the Amenity Facility, may be required to sign the Landlord's standard license agreement, waiver agreement and all keys, access cards, or fobs to such Amenity Facility shall be issued as outlined in such Rules and Regulations.

18. Access During Health Emergency

During a Health Emergency, the Landlord shall also be entitled to specify specific modes of ingress and egress from and to the Building for tenants generally, or for specific tenants, occupants or invitees who may have a heightened risk of either exposure to a health threat or a heightened risk of transfer of unhealthy condition to other tenants, invitees or visitors in the Building.

19. Disclosure by Tenant

The Tenant shall, immediately upon becoming aware of same, inform the Landlord of any outbreak of an infectious disease amongst its employees where such outbreak may impact the health and/or safety of other tenants in the Building or lead to a Health Emergency.

20. Health Emergency Drills

The Tenant shall participate in any Health Emergency drill that the Landlord shall choose to implement acting reasonably, in preparation for a Health Emergency.

SCHEDULE D INDEMNITY AGREEMENT

This Agreement is made the <*> day of <*>, 20<*>,

Between:

[name of Indemnifier],
 (the "Indemnifier")

- and

[name of Landlord],
 (the "Landlord").

WHEREAS:

The Landlord is the owner of the Lands and Premises Building known municipally as <*> [address of Building Project] (collectively, the "Building Project"); and

The Indemnifier and <*> [name of Tenant] (the "Tenant") have requested the Landlord to enter into a lease (the "Lease") dated <*>, 20<*> [date of Lease] between the Landlord, as landlord, and the Tenant, as tenant, relating to premises in the BuildingProject and the Landlord has agreed to do so only if the Indemnifier executes and delivers this Agreement in favour of the Landlord;

NOW THEREFORE for good and valuable consideration (the receipt and sufficiency of which are acknowledged by the Indemnifier), the Indemnifier agrees with the Landlord as follows:

- 1. The Indemnifier covenants with the Landlord that the Tenant will pay all Basic Rent, Additional Rent and other amounts payable under the Lease on the days and at the times and in the manner provided in the Lease, and will observe each and every covenant, proviso, condition, agreement and obligation contained in the Lease on the part of the Tenant to be performed and observed, and that if any default is made by the Tenant, whether in payment of monies or performance of obligations, the Indemnifier shall forthwith on demand pay to the Landlord such monies and perform such obligations and pay any and all damages resulting from any non-payment or non-performance.
- 2. The Indemnifier shall be jointly and severally liable with the Tenant for all of the Tenant's obligations under the Lease, as if it were separately named as a tenant under the Lease.
- 3. This Indemnity is absolute and unconditional and the obligations of the Indemnifier and the rights of the Landlord hereunder shall not be affected or in any way prejudiced or impaired by: (a) any neglect or forbearance by the Landlord in obtaining payment of Basic Rent, Additional Rent or other amounts or of enforcing the provisions of the Lease or the obligations of the Tenant or any waiver or failure to enforce any provision of this Agreement by the Landlord; (b) any extensions of time or other indulgences given by the Landlord to the Tenant; (c) any amendment of the Lease or other dealing between the Landlord and the Tenant with or without Notice to the

70

Indemnifier; (d) any Transfer by the Tenant (with or without the Landlord's consent); or (e) any other act or failure to act by the Landlord which would release, discharge or affect the obligations of the Indemnifier if it were a mere surety, with the intent that the obligations of the Indemnifier shall continue and shall not be released, discharged or reduced or in any way impaired until such time as all of the obligations of the Tenant under the Lease, now existing or to arise at any time in the future, have been fully performed and satisfied.

- 4. The Indemnifier expressly waives Notice of the acceptance of this Agreement and all Notice of non-performance, non-payment or non-observance on the part of the Tenant of the terms, covenants and conditions in the Lease. Without limiting the generality of the foregoing, any Notice which the Landlord desires to give to the Indemnifier shall be sufficiently given if personally delivered, delivered by courier or mailed by registered prepaid post, to the Indemnifier at the Premises, and every such Notice is deemed to have been given upon the day it was delivered, or if mailed, on the fifth day following the date of such mailing. The Indemnifier may designate by Notice in writing to the Landlord a substitute address for that set forth above. If two or more Persons are named as Indemnifier, such Notice given hereunder or under the Lease shall be sufficiently given if delivered or mailed in the foregoing manner to any one of such Persons.
- 5. The obligations of the Indemnifier under this Agreement shall not be released, discharged or affected by the bankruptcy or insolvency of the Tenant or any proposal made by it to its creditors or any repudiation of the Lease pursuant to the *Bankruptcy and Insolvency Act*, S.C. 1992, or any successor or similar legislation, or any disclaimer by any trustee in bankruptcy of the Tenant or by the Tenant ceasing to exist (whether by winding-up, forfeiture, cancellation or surrender of charter, or any other circumstance) or by any event terminating the Lease including a re-entry or termination. If the Lease is terminated prior to the end of its Term, except by surrender duly accepted by the Landlord, then, at the option of the Landlord, the Indemnifier shall execute a new lease of the Premises between the Landlord as landlord and the Indemnifier as tenant for a term equal in duration to the residue of the term remaining unexpired at the date of such termination, and in all other respects upon the same terms and conditions as are set forth in the Lease.
- 6. The Landlord shall not be obliged to exercise its remedies against the Tenant or any other Person or against the Premises or to exhaust any security given by the Tenant before demanding payment of monies or performance of covenants by the Indemnifier.
- 7. The Indemnifier's obligations under this Agreement bind the Indemnifier and its legal representatives, heirs, executors, administrators, successors and assigns (as the case may be) and may be assigned by the Landlord, and will benefit and be enforceable by the successors and assigns of the Landlord, and all parties who for the time being have the status of Landlord under the Lease, whether or not such parties receive a specific assignment of the Lease or of the Indemnifier's obligations, and whether or not Notice of any assignment or change in ownership of the Premises or any Project of which the Premises forms a part is given to the Indemnifier.
- 8. The grammatical changes required to make the provisions of this Agreement apply in the plural sense where the Indemnifier comprises more than one Person and to corporations, firms, partnerships, or individuals male or female, will be assumed as though in each case fully

expressed, and if the Indemnifier consists of more than one Person, the obligations of the Indemnifier shall be deemed to be joint and several obligations of each such Person. This Agreement shall be construed in accordance with the Applicable Laws.

- 9. The Indemnifier acknowledges receipt of a copy of the Lease and covenants, represents and warrants that it has full power, capacity and authority to enter into this Agreement and to perform its obligations hereunder. No modification of this Agreement shall be effective unless it is in writing and is executed by both the Indemnifier and the Landlord.
- 10. Whenever any reference is made in this Agreement to the Lease or the obligations of the Tenant under the Lease, such reference shall be deemed to include any and all agreements and instruments executed by the Tenant in connection with the Lease or pursuant to the Lease and which relate to the Premises. Any capitalized word or phrase used in and not defined in this Agreement shall have the meaning given to it in the Lease.

| IN WITNESS WHERI | COF the Indemnifier has executed this Agreement under seal. |
|---------------------------|---|
| | <*> (Indemnifier) |
| | Per: |
| | Name: |
| | Title: |
| | e /s |
| | Per: |
| | Name: |
| | Title: |
| | I/We have authority to bind the Corporation. |
| [If the Indemnifier is no | ot a corporation then replace its signing line with the following:] |
| Signed, sealed and deliv | ered in the presence of: |
| | 1/s |
| Name of Witness: | [Individual's Legal Name] |

SCHEDULE E ENVIRONMENTAL SUSTAINABILITY MANAGEMENT PLAN

SECTION 1 - ENVIRONMENTAL - SUSTAINABILITY OBJECTIVES

1.1 Context

The provisions of this Environmental Sustainability Management Plan have been designed to encourage and promote the implementation of certain environmental sustainability objectives on the part of each of the Landlord and the Tenant.

[OPTION 1: The provisions of this EnvironmentSustainability Management Plan shall form part of this Lease and comprise a covenant on the part of the Landlord or the Tenant, as the case may be, respectively.]

[OPTION 2: A breach by either the Landlord or the Tenant of any of the provisions of this EnvironmentalSustainability Management Plan on the part of either the Landlord or the Tenant to be observed or performed, as the case may be, shall not constitute a default under this Lease, but the party committing such breach agrees, to the extent possible under the circumstances, to use commercially reasonable efforts to cooperate with the other party to remedy such breach. In addition to the foregoing, the Tenant and the Landlord agree to constructively consult with each other on: (i) enhancements that may achieve the EnvironmentalSustainability Objectives and the Landlord and Tenant shall consider undertaking any such enhancements; and (ii) issues, events and circumstances likely to detract from achieving the EnvironmentalSustainability Objectives.]

The Landlord and Tenant shall each provide a point of contact to discuss issues that pertain to the Sustainability Management Plan.⁹

| Tenant sustainability contact: |
|----------------------------------|
| Email: |
| Phone: |
| Landlord sustainability contact: |
| Email: |
| Phone: |

The Tenant sustainability contact (or at Tenant's option, another staff representative) shall be made available to participate, from time to time, in the Landlord's information and training sessions relating to the Landlord's initiatives for improving the operational sustainability of the Project.

 $[\]frac{9}{https://www.greenleaseleaders.com/wp-content/uploads/2021/02/Green-Lease-Leaders-Landlords-Reference-Guide-011521.pdf}$

1.2 General Objectives

- (a) The Tenant acknowledges the Landlord's intention to operate the **BuildingProject** so as to provide for:
 - (i) a comfortable, productive and healthy indoor environment;

73

- (ii) reasonable measures being taken to clean and operate the Building to prevent the spread and transmission of any potential pathogens;
- (iii) reduced energy use and reduced production, both direct and indirect, of Greenhouse Gases;
- (iv) (iii) reduced use of potable water and (such that water quality continues to meet applicable standards such as Guidelines for Canadian Drinking Water Quality¹⁰), including by safely increasing the use of recycled water (including non-potable water, such as rainwater or alternative water systems) where appropriate;
- (v) (iv) the effective diversion of construction, demolition, and land-clearing waste from landfill and incineration disposal, and the recycling of tenant waste streams, and the reduction of materials purchased where feasible;
- (vi) the useprocurement of cleaningsustainable and healthy materials and products certified in accordance with EcoLogo^M (Canada), Green SealTM (United States) or equivalent, Forest Stewardship Council (FSC)TM, ENERGY STAR, or other appropriate standards;(vi) given the facilitation of alternate transportation options for individuals attending at the Building;(vii) product, and the avoidance of high-VOC emitting materials, furniture and improvements within the BuildingProject and individual tenant premises; and
- (vii) the facilitation of sustainable transportation options and infrastructure for individuals attending at the Project, including public transit, electric vehicles, and cycling;
- (viii) minimized light pollution from, and bird collisions with, the Building;
- (ix) effective management and reduction of, where relevant, any Hazardous Substances present in or on the Project, including, but not limited to asbestos, legionella and radon;
- (x) compliance with all applicable Environmental Laws;

¹⁰ https://www.canada.ca/en/health-canada/services/environmental-workplace-health/reports-publications/water-quality/guidelines-canadian-drinking-water-quality-summary-table.html

© Copyright, Real Property Association of Canada, 2009-, 2017, 2021

- (xi) supporting, improving, and/or maintaining the Resilience of the Premises and the Project at large to Natural Hazards; and
- (xii) (viii) [Optional] the achievement of such other more specific targets as may be set out in Section 1.3 below.
- (b) The Tenant also acknowledges that the Building currently has achieved or qualifies for the following accreditations, ratings or certifications: [NTD: choose as applicable]
 - (i) LEED® for NewBuilding Design & Construction and Major Renovations ("NCBD+C") certified [(silver, gold, platinum, as applicable)];
 - (ii) LEED® for Core and ShellInterior Design & Construction ("CSID+C") certified [(silver, gold, platinum, as applicable)];
 - (iii) LEED® for Existing Buildings: Building Operations and Maintenance ("EB:O&+M") certified [(silver, gold, platinum, as applicable)];
 - (iv) Building Research Establishment Environmental Assessment Method ("BREEAM") rating of ;
 - (iv) (v) ENERGY STAR rating of _____;
 - (v) BOMA BEST certified [(bronze, silver, gold, platinum, as applicable)];
 - (vi) WELL certified [(silver, gold, and platinum, as applicable)];
 - (vii) Fitwel certified [(1-star, 2-star, 3-star, as applicable)];
 - (viii) Rick Hansen Foundation (RHF) Accessibility Certified [(certified or gold certified, as applicable)];
 - (ix) Canada Green Building Council Zero Carbon Building (ZCB) Design or Performance Certification, as relevant, or other national or international standard for determining net zero in force or recommended for use within the Canadian commercial real estate industry; and/or
 - (vi) BOMA BESt rating of <*>.
 - (x) Other. [NTD: American Society of Heating, Refrigerating, and Air-Conditioning Engineers ("ASHRAE"), Passive House or other standard]

The Tenant agrees that the Landlord shall be entitled to operate, manage and maintain the Building so as to retain at least such level of accreditation, rating or certification. The Tenant agrees to also share any relevant data with the Landlord needed to retain at least such level of accreditation, rating or certification.

- (c) The Tenant acknowledges <u>that</u> the Landlord's <u>intention to may</u> operate the Building so as to achieve and retain <u>all</u>, some or any of the following building accreditations:
 - (i) a LEED[®] for Existing Buildings: Operations Building Operation and Maintenance ("EB:O&±M") certified standard [(silver, gold, platinum) as applicable];
 - (ii) a top quartile ranking under BOMA BESt; [(bronze, silver, gold, or platinum, as applicable)] certification under BOMA BEST;
 - (iii) a top quartile ranking under the ENERGY STAR program;
 - (iv) a WELL certification [(silver, gold, and platinum, as applicable)];
 - (v) a Fitwel certification [(1-star, 2-star, 3-star, as applicable)];
 - (vi) a Rick Hansen Foundation (RHF) Accessibility Certification [(certified or gold certified, as applicable)];
 - (vii) Canada Green Building Council Zero Carbon Building (ZCB) Performance
 Certification, or other national or international standard for determining net
 zero in force or recommended for use within the Canadian commercial real
 estate industry; and/or
 - (viii) (iv) other Other. [NTD: add any other specific accreditation, such as Green Building Initiative's Green Globes for Continual Improvement for Existing Buildings ("CIEB"), American Society of Heating, Refrigerating, and Air-Conditioning Engineers ("ASHRAE"), Passive House or other standard]
- (d) The Landlord shall be entitled from time to time during the Term, to seek such other and further building certifications as may be reasonably necessary, in the Landlord's sole opinion, to ensure the BuildingProject (or any part thereof) remains compliant with all Applicable Laws (including expected enhancements thereto), as well as certifications prevalent in the marketplace or necessary to attract leading tenants from time to time.

1.3 Specific Objectives

- (a) Notwithstanding the provisions of Section 1.2 in this Schedule "E" above, the Tenant acknowledges the Landlord's intention to achieve, and maintain, the following specific targets for the BuildingProject, by >> [NTD: target date if not today]:
 - (i) electricity use averaging not greater than

 of Rentable Area of the Building per year (kWh/sf/yr) for equivalent

kilowatt hours per square foot of Rentable Area where normalization of data has occurred];

- (i) (ii) natural gasenergy use intensity (electricity and fuel consumption averaging) not greater than <*> eubic metresequivalent kilowatt hours per square foot of RentableGross Floor Area of the Building per year (M³ekWh/sf/yr) [or equivalent eubic metreskilowatt hours per square foot of RentableGross Floor Area where normalization of data has occurred]¹¹;
- (ii) water consumption levels averaging not greater than <*> litres per square foot of Rentable Gross Floor Area of the Building per year (l/sf/yr) [or equivalent litres per square foot of Rentable Gross Floor Area where normalization of data has occurred] and such that the minimum usage required to maintain the water quality for the Project is in accordance with applicable standards such as the Guidelines for Canadian Drinking Water Quality (13);
- (iii) (iv)-a waste diversion rate not less than <*>% per year; and
- (iv) indoor CO₂ levels compared to outdoor CO₂ levels of not greater than Parts Per Million ("PPM") measured in accordance with the ASHRAE Standard 62.1 200462.1-2019 (or later) or equivalent standard as it may be amended or replaced from time to time.
- (b) The Landlord shall be entitled, acting reasonably, at any time and from time to time, to adjust the foregoing specific targets for the Building based on the type and intensity of space usage within the Building, having regard to the then current tenant base for the Building, the energy or other resource consumption profile of the Tenant, or to "normalize" the foregoing specific targets for the Building having regard to the then tenant base of the Building, the energy or other resource consumption profile of the Tenant, and the change in use or energy consumption for various parts of the Building, including underground parking areas (if any), food courts (if any) or other Common Areas and Facilities.
- (c) In addition, the Landlord shall be entitled to modify the foregoing specific targets to accord such targets with the standards that may be established pursuant to a third party certification or rating scheme such as LEED EB:O&±M, ENERGY STAR, BOMA BEStBEST, or any specific building labelling scheme that may be promulgated in Canada either on a voluntary or mandatory basis, from time to time.

¹¹ https://www.realpac.ca/page/NEUINWUI

¹² https://www.realpac.ca/page/NEUINWUI

¹³ https://www.canada.ca/en/health-canada/services/environmental-workplace-health/reports-publications/water-quality/guidelines-canadian-drinking-water-quality-summary-table.html

1.4 Regulatory Standards

Notwithstanding the provisions of Section 1.2 and 1.3 herein, in the event that any governmental authority imposes a resource reduction target on the Building for any utility—or, resource, or emission otherwise than as set out in Sections 1.2 and 1.3 above, then the Environmental Sustainability Objectives shall be deemed to have been amended so as to stipulate such resource reduction target and all changes required to be made by the Landlord to the Environmental Sustainability Management Plan, or which are necessitated as a result of such mandatory resource reduction target, shall be deemed to be included and permitted, as the case may be, pursuant to the provisions of this Section and this Lease.

1.5 Carbon Offset Credits and Carbon Offset Costs

- (a) The Landlord shall be entitled to all Carbon Offset Credits that may be created, credited or recoverable as a result of activities conducted within the Premises or the BuildingProject (or any part thereof), excluding Carbon Offset Credits to which the Tenant is entitled in accordance with Applicable LawLaws. The Landlord shall be entitled to allocate, acting reasonably, to the tenants of the BuildingProject, Carbon Offset Credits (net of all costs of aggregating, auditing and certifying same, not otherwise included in Operating Costs) created with the participation of the Tenant and/or other tenants inof the BuildingProject.
- (b) Where, as a result of energy consumption in the Building, the Landlord is required to incur a Carbon Offset Cost for the Project, such Carbon Offset Cost shall be included in Operating Costs and recoverable pursuant to the provisions of Section 6.5 of this Lease.

1.6 Determination of Compliance

Any issue in respect of the compliance of either party with the general objectives set out in Section 1.2 above or the specific objectives set out in Section 1.3 above shall be determined by an Expert as appointed by the Landlord, and the provisions of Section 16.6 of thethis Lease to which this Schedule forms a part, shall apply to such Expert's determination. Such Expert shall advise, in respect of any question pertaining to the achievement of a specific objective of the BuildingProject or the Premises as to why the BuildingProject or the Premises, as the case may be, does not appear to be or have achieved such objective or target, the Expert's perspective on the allocation of responsibility for such non-performance, and recommendations for improvement or ways in which the prescribed objective or target could be achieved.

SECTION 2 - ENVIRONMENTAL SUSTAINABILITY MANAGEMENT PLAN IMPLEMENTATION

2.1 The Tenant agrees to conduct its operations in the <u>BuildingProject</u> and within the Premises in accordance with the following provisions:

- (a) Comfortable, Healthy and Productive Indoor Environment
 - (i) The Landlord shall be entitled at any time and from time to time to undertake Greenhouse Gas production Indoor Air Quality (IAQ) monitoring and testing, including testing within the Premises, on reasonable Notice to the Tenant and accompanied by a representative of the Tenant if required, which representative Tenant agrees to make available.
 - (ii) The Tenant shall ensure that all work done within the Premises by the Tenant or its representatives shall be undertaken in accordance herewith and with the Tenant Construction Manual. Notwithstanding the foregoing, the Tenant shall specify that all paints, sealants and adhesives used or to be used within the Premises meet EcoLogo^M, Green SealTM, South Coast Air Quality Management District ("SCAQMD") regulations, MPI Green PerformanceTM Standards or equivalent so as to ensure no or low emissions of VOCs within the BuildingProject. Landlord may from time to time conduct tests to measure VOCs within the Premises.
 - (iii) The Tenant shall have regard to the Tenant Procurement Guidelines in procuring furniture, fixtures, materials, supplies and equipment to be brought into the Premises.
 - (iv) Should the Tenant be permitted to undertake its own cleaning of, or within, the Premises, the Tenant shall require that in any cleaning contracts granted directly by it, the cleaning contractor shall use cleaning products certified in accordance with EcoLogo^M, Green SealTM or equivalent. The Landlord shall reserve the right to approve, acting reasonably, any such Tenant cleaning contracts, but without liability. The Tenant shall ensure that any cleaning contracts entered into by it require the cleaning contractor to comply with elements of the Environmental Sustainability Management Plan applicable to it. Particularly, any cleaning contracts letentered into by the Tenant in respect of specialized green facilities, such as waterless urinals, shall ensure the cleaning contractor properly understands and is trained on the maintenance of such specialized green facilities.
 - (v) At the Tenant's sole cost and expense, and subject to the approval of the Landlord acting reasonably, the Landlord agrees to purge Building air during a Tenant move in to minimize off-gassing of wallpaper, carpet and furniture glues and dyes in accordance with the Sheet Metal and Air Conditioning National Contractors Association ("SMACNA") Indoor Air Quality (IAQ) Guidelines for Occupied Buildings Under Construction, 2nd edition ANSI/SMACNA, 2008, or equivalent.
 - (vi) The Landlord shall operate and maintain and operate all Building Systems in accordance with relevant industry standards. HVAC systems shall be operated and maintained in accordance with ASHRAE 62.1-2019, Section

8 (or later), or equivalent standard. Building water systems shall be operated and maintained in accordance with ASHRAE 188-2018 and/or CDC Toolkit: Developing a Water Management Program to Reduce Legionella Growth and Spread in Buildings, or equivalent standard.

- (vii) OPTIONAL: [In undertaking any work within the Premises, the Tenant shall ensure that
 - (A) its contractors follow the air quality approach of the Sheet Metal and Air Conditioning National Contractors Association ("SMACNA") Indoor Air Quality Guideline(IAQ) Guidelines for occupied buildings under construction, 1995 (Chapter 3)Occupied Buildings Under Construction, 2nd edition ANSI/SMACNA, 2008 (or later) or Canadian equivalent standard as it may be amended or replaced from time to time.
 - (B) filtration media for any air filtration required has a Minimum Efficiency Reporting Value ("MERV") of 813 as determined by ASHRAE 52.2 19992017, or Canadian equivalent or as amended from time to time and ensure that any such filtration media are replaced after construction and immediately prior to occupancy with filtration media having a MERV of 13.
 - (C) all paints and coatings achieve Green Seal's standard GS-11 or Canadian equivalent requirement, all carpet meets the Carpet and Rug Institute ("CRI") Green Label Plus Carpet Testing Program requirements, and any carpet cushion meets the requirements of the CRI Green Label Testing Program or Canadian equivalent.
 - (D) sustainable cleaning chemicals shall meet the Green Seal GS-37 standard or Canadian equivalent and carpet care products shall meet the requirements of GS-37 or Canadian equivalent and/or EcoLogo CCD-148UL 2795 Carpet and Upholstery Cleaners.
 - (E) floor finishers and strippers shall adhere to Green Seal's standard GS-40 or Canadian equivalent and/or EcoLogo CCD-147.(F)UL 2777 Hard Floor Care Products contractors shall reduce connected lighting power density below that allowed by ANSI/ASHRAE/IESNAIES Standard 90.1-200490.1-2019 (or later) by a minimum of 15%].
- (b) Reduce Indirect and Direct Energy Consumption and Greenhouse Gas Emissions
 - (i) The Tenant agrees to the installation of electricity smart meters in respect of the Tenant's consumption of electricity within the Premises, at the Tenant's sole cost and expense, payable as Additional Rent under this

80

<u>Lease</u>. The Landlord shall have access to this aggregated data in accordance with Section 7.2(d) of this Lease.

- (ii) The Tenant shall take reasonable steps to minimize its electrical consumption within the Premises such as, by way of example only, adopting conservation practices (e.g.including reducing its use of lighting where unnecessary); the use of ENERGY STAR equipment; the types of lighting, lighting switches, sensors and zones as may be specified in the Tenant Construction Manual; and using the types of equipment suggested in the Tenant Procurement Guidelines or the REALpac Green Lease Guide for Commercial Office Tenants.
- (iii) The Landlord shall be entitled at any time or from time to time to acquire (A) all or part of the power for its Common AreaAreas and Facilities; or (B) shared electrical power from sources with low Greenhouse Gas emissions. In addition to the foregoing, where it is considered feasible to do so, the Landlord may install onsite generation capacity either to reduce peak load or to supplement base load requirements for the BuildingProject from time to time, and any incremental cost in so doing shall be included in Operating Costs. Without limiting the generality of the foregoing, the Tenant shall, where available on commercially reasonable terms, ensure that equipment purchased for the Premises is ENERGY STAR certified where available.
- (iv) The Tenant shall purchase energy from on-site renewables if offered by the Landlord. The Landlord shall install, own, and maintain the on-site generation and sell power directly to the Tenants at a cost not exceeding that of local gas and electricity utilities as applicable. ¹⁴ If on-site renewables are not applicable, the Tenant shall be entitled at any time or from time to time to specify in writing that it wishes to have its electrical power consumption sourced or offset from renewable energy sources, and if it shall elect to do so, the cost of same shall be at the Tenant's sole cost and expense, either payable directly by it to the supplier so chosen, or recoverable by the Landlord if paid by the Landlord as Additional Rent.
- (v) The Landlord shall be entitled to benchmark itself against any building rating system for electrical, natural gas, water or other resource consumption.
- (vi) The Landlord shall operate <u>Buildingthe</u> Common Areas and Facilities in accordance with, and use its reasonable efforts to cause other tenants to operate in conformity with, the <u>Environmental Objectives.Sustainability Objectives.</u>

© Copyright, Real Property Association of Canada, 2009-, 2017, 2021

_

¹⁴ https://www.greenleaseleaders.com/wp-content/uploads/2021/02/Green-Lease-Leaders-Landlords-Reference-Guide-011521.pdf

- when finishing out the Project, including the Premises and Common Areas and Facilities, the Landlord will install the lighting systems with daylight zones in atriums or within 15 feet of windows and skylights where daylight can contribute to energy savings. Daylight zones will be required per International Energy Conservation Code (IECC) 2021, or equivalent standard, and will be either manual off via switches or automated via daylight sensors. Lighting controls (including occupancy sensors and scheduling controls) will be provided for all lighting equipment according to IECC 2021, or equivalent standard. The occupancy sensors will function as either manual ON (vacancy mode) or automatically turn ON not more than 50% of the lighting power. The occupancy sensors will then automatically turn lights OFF when no human presence is detected for a specified amount of time.
- (viii) The Landlord shall provide the Tenant with the Building's ENERGY STAR score annually. To the extent the Tenant obtains electricity independently of the Building, the Tenant shall give the Landlord access to the Tenant's data on energy use for inclusion in the Landlord's annual reports, ENERGY STAR annual rating and similar purposes.¹⁵
- (ix) The Tenant shall be required to submit on a(n) [monthly, quarterly, annual] basis to the Landlord energy and water consumption data, including total usage and total charges as they appear on the Tenant's electric, gas, water, and other utility bills, in a format deemed reasonably acceptable by the Landlord. 16
- (x) The Tenant shall be required to submit on a(n) [monthly, quarterly, annual] basis to the Landlord purchase data of green power, RECs, and carbon offsets purchased for the Premises, in a format deemed reasonably acceptable by the Landlord.
- (xi) All Alterations will be performed in accordance with the Landlord's sustainability practices that the Tenant has accepted as part of this Lease, namely the Premises fit-out must meet Environmental Protection Agency's ENERGY STAR Tenant Space criteria or equivalent. 17
- (xii) (vii) [Optional: The Tenant agrees to limit annual average consumption of electricity within its the Premises to <*> watts kilowatt-hours per square foot, exclusive of Building standard lighting and specific equipment which may be approved by the Landlord in writing from time to time.]

¹⁵ https://www.gsa.gov/cdnstatic/GBAC HP Leasing Criteria - FINAL.pdf

¹⁶ https://www.imt.org/wp-content/uploads/2018/02/Making Efficiency Work for You.pdf

¹⁷ https://www.greenleaseleaders.com/wp-content/uploads/2021/02/Green-Lease-Leaders-Landlords-Reference-Guide-011521.pdf

(c) Net Zero Carbon Building Goal

(i) The Landlord may design and/or operate the Project to be certified as a Zero Carbon Building ("ZCB") in accordance with the Canada Green Building Council Zero Carbon Building Design Standard Version 2 (or later), the Canada Green Building Council Zero Carbon Building Performance Standard Version 2 (or later) and/or other national or international standard for determining net zero in force or recommended for use within the Canadian commercial real estate industry. The Tenant shall work with the Landlord to meet the requirements set out in the applicable standard to achieve and/or maintain ZCB certification.

(d) (e) Reduce Water Consumption

- (i) The Tenant agrees to the installation of water meters or check meters in respect of the Tenant's consumption of water, at the Tenant's sole cost and expense, payable as Additional Rent under this Lease. The Landlord shall have access to this aggregated data in accordance with Section 7.2(d) of this Lease.
- (ii) Water quality shall continue to meet applicable standards such as Guidelines for Canadian Drinking Water Quality¹⁸.
- (iii) Where potable water usage is not a necessity, the Tenant acknowledges and consents to the use of treated recycled or treated natural water in washrooms and in other applications within and around the BuildingProject.
- (iv) (iii) The Tenant consents to rainwater collection, treatment and reuse by the Landlord and wastewater collection, treatment and reuse by the Landlord from time to time. The Tenant consents to the use of water-saving appliances, such as waterless urinals, and other equipment as may be otherwise consistent with the EnvironmentalSustainability Objectives.
- (v) (iv) The Tenant shall ensure that any fixtures purchased for the Premises shall be certified by the United States' Environmental Protection Agency's Water Sense Program or a Canadian equivalent, if any, and follow the Energy Policy Act of 1992 (or later amended), for water fixture performance requirements, or Canadian equivalent, as referenced in relevant LEED Reference Guides.
- (vi) (v) [Optional: The Tenant agrees to limit annual average consumption of water within its Premises to *> litres per square foot of Rentable Area of the Premises exclusive of consumption within Building standard washrooms.]

© Copyright, Real Property Association of Canada, 2009-, 2017, 2021

¹⁸ https://www.canada.ca/en/health-canada/services/environmental-workplace-health/reports-publications/water-quality/guidelines-canadian-drinking-water-quality-summary-table.html

(e) (d) Recycled and/or Sustainable Materials Usage and Waste Diversion

- (i) The Tenant shall be entitled to use recycled and/or sustainable materials in its Leasehold Improvements and Alterations if so permitted either pursuant to the Tenant Construction Manual, or as may be consented to by the Landlord, acting reasonably. The Tenant agrees to consider locally extracted, harvested, sourced, and manufactured materials where possible in the completion of Leasehold Improvements Alterations, consistent with the terms as set out in the Tenant Construction Manual and the Tenant Procurement Guidelines.
- (ii) Tenant shall be entitled to use recycled <u>and/or sustainable</u> furniture, fixtures and equipment in the Premises to the extent consistent with the <u>EnvironmentalSustainability</u> Objectives and the Tenant Procurement Guide.
- (iii) The Tenant agrees to recycle or reuse or cause its contractor to recycle or reuse as much as possible any waste created in the demolition of existing Leasehold Improvements or Alterations within the Premises so as to minimize the amount of waste ending in landfill. The Landlord reserves the right to monitor and measure the amount of waste leaving the Building from the Premises and going to landfill from time to time. If available, the Landlord agrees to provide to the Tenant a staging area for the sorting and recycling of materials during construction.
- (iv) The Landlord may refuse to collect or accept from the Tenant's-Premises waste that is not appropriately sorted into the appropriate recycling container.
- (v) The Landlord shall have access to aggregated waste collection and hauling data from the Premises in accordance with Section 7.2(d) of this Lease. In addition, upon the Landlord's demand, acting reasonably, the Tenant accepts to provide the Landlord with reports regarding its volume of internal recycling (including, among others, secure shredding totals).

(f) (e) Tenant Certifications

- (i) The Landlord will use commercially reasonable efforts to co-operate with the Tenant, at the Tenant's sole cost, in the certification of the Premises pursuant to any rating scheme, such as ASHRAE standard 189.1189.12017, or later, LEED-CIID+C standard (as specified by the U.S. Green Building Council until adopted by the Canada Green Building Council) or equivalent standard as the Landlord may agree to, acting reasonably.
- (ii) The Tenant agrees to provide all reasonable information required by the Landlord consistent with the accreditation or certifications contained in the EnvironmentSustainability Management Plan, in a form acceptable to the Landlord, acting reasonably, within 10ten Business Days of request.
- (g) (f) External Environment

- (i) The Tenant shall work with the Landlord to minimize light pollution and collision of birds with the Building by following the practices outlined in LEED v4.1 (or later) or equivalent.
- (ii) [OPTIONAL: The Tenant shall ensure that any exterior work undertaken by it shall prevent loss of soil during construction, shall protect any soil stockpiled for re-use, shall minimize soil erosion from wind and water and shall prevent dust and air pollution due to wind blowing over any such soil or other construction materials.]

SECTION 3 - ENVIRONMENTAL ASSESSMENT AND REPORTING

3.1 The Landlord and Tenant, acting reasonably and in good faith, agree to cooperate from time to time in determining compliance with the EnvironmentalSustainability Objectives as set out in Section 1 herein and in refining such EnvironmentalSustainability Objectives from time to time. The Landlord and the Tenant agree to meet at least EnvironmentalSustainability Objectives for the BuildingProject and the Premises and any further steps that could be taken to achieve the EnvironmentalSustainability Objectives.

SECTION 4 – BUILDING RESILIENCE PLAN

4.1 Context

The Building Resilience Plan exists as part of the Sustainability Management Plan as a way for the Project to respond to and mitigate possible Natural Hazards and Hazardous Events that may be experienced during the Term of this Lease.

4.2 Building Resilience Plan Implementation

The Tenant agrees to implement its own, or follow the Landlord's directive, to support, improve, and/or maintain the Project's Resilience. The Landlord and Tenant shall agree to which Natural Hazards are relevant to the Project and the adaptation measures that should be covered in the Building Resilience Plan. For example, if flooding is a relevant Natural Hazard, appropriate adaptation measures may include creating an emergency preparedness and response plan, conducting flood event procedures and drills annually, establishing an emergency stakeholder communication channel (such as, by way of example, mass notification systems and social media channels) in which contact details are regularly updated, and having critical equipment and supplies on hand. Adaptation measures may include retrofits to the Premises where possible. For example, to improve flood resilience, this may include elevating and flood-proofing critical equipment (including HVAC equipment; electrical transformers, switchgear and service panels; communication systems; server rooms; high-voltage and telecommunication pull rooms; and isolating electrical circuits). ¹⁹

© Copyright, Real Property Association of Canada, 2009-, 2017, 2021

.

 $[\]frac{^{19}\ https://www.intactcentreclimateadaptation.ca/wp-content/uploads/2019/10/Ahead-of-the-\underline{Storm-1.pdf}$

[NTD: Draftsperson to insert Natural Hazard adaptation measures relevant to the Building as agreed upon by the Landlord and Tenant]

5954546.3